



Article

Re-Examining Confidence Intervals for Ratios of Parameters

Zaka Ratsimalahelo

CRESE (UR 3190), Université Marie et Louis Pasteur, F-25000 Besançon, France; zaka.ratsimalahelo@univ-fcomte.fr

Abstract

This paper considers the problem of constructing confidence intervals (CIs) for nonlinear functions of parameters, particularly ratios of parameters a common issue in econometrics and statistics. Classical CIs (such as the Delta method and the Fieller method) often fail in small samples due to biased parameter estimators and skewed distributions. We extended the Delta method using the Edgeworth expansion to correct for skewness due to estimated parameters having non-normal and asymmetric distributions. The resulting biascorrected confidence intervals are easy to compute and have a good coverage probability that converges to the nominal level at a rate of $O(n^{-1/2})$ where n is the sample size. We also propose bias-corrected estimators based on second-order Taylor expansions, aligning with the "almost unbiased ratio estimator". We then correct the CIs according to the Delta method and the Edgeworth expansion. Thus, our new methods for constructing confidence intervals account for both the bias and the skewness of the distribution of the nonlinear functions of parameters. We conduct a simulation study to compare the confidence intervals of our new methods with the two classical methods. The methods evaluated include Fieller's interval, Delta with and without the bias correction interval, and Edgeworth expansion with and without the bias correction interval. The results show that our new methods with bias correction generally have good performance in terms of controlling the coverage probabilities and average length intervals. They should be recommended for constructing confidence intervals for nonlinear functions of estimated parameters.

Keywords: ratio parameters; confidence intervals; bias; almost unbiased ratio estimator; edgeworth expansion; Cornish–Fisher expansion; Delta method; Fieller method; coverage probability

JEL Classification: C12; C13



Academic Editor: Stephen G. Hall

Received: 24 June 2025 Revised: 28 August 2025 Accepted: 11 September 2025 Published: 20 September 2025

Citation: Ratsimalahelo, Z. (2025). Re-Examining Confidence Intervals for Ratios of Parameters. *Econometrics*, 13(3), 37. https://doi.org/10.3390/ econometrics13030037

Copyright: © 2025 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/).

1. Introduction

Many econometric and statistical applications are interested in tests of the nonlinear functions of the parameters, which can be expressed as the ratio of two unknown parameters including the ratio of regression coefficients, the ratio of the two linear functions such as the ratio of affine transformations of random variables and generally the ratio of the two nonlinear functions.

A non-exhaustive list of examples of econometric models where inferences for the ratio of parameters are used as follows: the long-run elasticities and flexibilities in dynamic models (Li & Maddala, 1999; Dorfman et al., 1990; Bernard et al., 2007; J. G. Hirschberg et al., 2008); the willingness to pay value, i.e., the maximum price an agent would pay to obtain an improvement in a particular attribute of a desired good or service (Lye &

Hirschberg, 2018); the turning point in a quadratic specification model where the estimated relationship is either a U-shaped or an inverted U-shaped curve, for example, Kuznet and Beveridge curves, in applications to dynamic panel data (Bernard et al., 2019; Lye & Hirschberg, 2018); the determination of the non-accelerating inflation rate of unemployment (NAIRU); for example, a Phillips curve (Staiger et al., 1997; J. Hirschberg & Lye, 2010c; Lye & Hirschberg, 2018); the structural parameter in an exactly identified system of equations as estimated by the two-stage least squares method (J. Hirschberg & Lye, 2017; Lye & Hirschberg, 2018; Andrews et al., 2019); the notion of weak instruments in econometric models (Woglom, 2001); inequality indices (Dufour et al., 2018, 2024); structural impulse responses (Olea et al., 2021). Lye and Hirschberg (2018) give some other examples of econometric models.

Other examples of statistical applications, include cost-effectiveness analysis (Briggs & Fenn, 1998); and the comparison of health outcomes across spatial domains (Beyene & Moineddin, 2005); bioequivalence assessment, dose–response analysis (Sitter & Wu, 1993; Faraggi et al., 2003; Y. Wang et al., 2015); the estimation of willingness to pay (Leitner, 2024) And other statistical applications, P. Wang et al. (2021); Raghav et al. (2025).

However, the statistical properties of the ratio of parameters can be problematic because the analytical expressions of the moments are generally not available, e.g., the ratio of asymptotically normally distributed random variables is a non-central Cauchy distribution. Moreover, if the denominator of the ratio is not significantly different from zero, the probability distribution of the ratio shows unusual behavior, and the confidence intervals are unbounded. Another problem worth highlighting is the bias of the estimator in a finite sample when studying a nonlinear function of parameters.

To test the null hypothesis of the nonlinear functions of parameters, which can be expressed as the ratio of two unknown parameters, we use confidence intervals (CIs). The two widely used approaches for constructing CIs are the Fieller method and the Delta method. The advantage of these methods is that they can be implemented in any context and are easy to compute, they do not require the use of intensive calculation and sampling strategies as would be needed when using a Bootstrap or Bayesian method (J. Hirschberg & Lye, 2010c; J. Hirschberg & Lye, 2017; Lye & Hirschberg, 2018).

Fieller (1954) proposed a method to derive the confidence interval (CI) of the ratio of two random variables. In Fieller's method, it assumes that both the numerator and the denominator of the ratio follow normal distribution. The method is based on the inversion of the pivotal *t*-statistic, which gives an exact CI for achieving the required coverage probability. The Fieller's CI is asymmetric around the ratio estimate, which is a good property, as it can be reflected in the skewness of the small sample distribution of the ratio. However, if the denominator of the ratio is not significantly different from zero, Fieller's CI will be unbounded, being either the entire real line or the union of two disconnected infinite intervals. It has a positive probability of producing CI with infinite length. Furthermore, Fieller's interval requires finding roots of a quadratic equation and these can be imaginary. In addition, if this quadratic equation has one root, the confidence interval will be half-open.

The Delta method is based on the first-order Taylor expansion by considering non-linear functions of parameters. By assuming asymptotic normality in large samples, this method produces a symmetric and bounded CI, unlike the Fieller method. However, the Delta method often has an inaccurate coverage probability (Dufour, 1997) and unbalanced tail errors even at moderate sample sizes (J. Hirschberg & Lye, 2010c). A geometric interpretation of the Fieller and Delta methods can be found in von Luxburg and Franz (2009); J. Hirschberg and Lye (2010a, 2010b). According to J. Hirschberg and Lye (2010c), if the true value of the ratio has the same sign as the correlation coefficient between the numerator

Econometrics **2025**, 13, 37 3 of 27

and the denominator then the Delta and Fieller intervals may be very similar even if the denominator has a high variance. However, if the signs are opposite and the precision of the denominator is low, then the Delta method has poorer performance.

Moreover, there are two potential problems with these Fieller and Delta methods; first, the parameter *estimator* is biased for nonlinear parameter functions. Second, the estimated parameters have non-normal and asymmetric distribution. Thus, the variance of the estimated parameters is not useful in constructing confidence intervals, Dorfman et al. (1990); Li and Maddala (1999).

In order to overcome the *limitations* of the previous methods, some numerical procedures have been proposed in the literature such as the parametric bootstrap method and the nonparametric bootstrap method (bootstrap standard, bootstrap t-statistic; bootstrap percentile, bootstrap bias-corrected, bootstrap bias-corrected and accelerated) see Krinsky and Robb (1986); Dorfman et al. (1990); Li and Maddala (1999), among others. The CIs obtained from these iterative procedures are bounded and are more computationally intensive.

Dorfman et al. (1990) compared the Delta and Fieller methods and three types of the single bootstrap and found that the bootstrap did not achieve nominal coverage and that all methods performed reasonably well.

The bootstrap percentile-t and the Delta methods confidence intervals are very close to each other in many cases in terms of the length of the confidence intervals, Li and Maddala (1999).

It should be noted that all the previous methods do not take into account the bias of the estimator, which should be a prerequisite for constructing a reliable confidence interval.

In this regard, the paper has five main contributions. First, we propose a novel analytical approach that modifies the Delta method to reduce the effect of skewness. This method is based on the Edgeworth expansion (Hall, 1992b). We then propose an easy to compute confidence interval for the ratio of parameters and the interval has the coverage probability converging to the nominal level at a rate of $O(n^{-1/2})$ where n is the sample size. Second, the source of potential bias is due to the nonlinearity of the ratio $\hat{\theta} = \hat{\theta}_1 / \hat{\theta}_2$ in terms of θ_1 and θ_2 . It is well known that even when exact unbiased estimators of θ_1 and θ_2 are available, the ratio estimator θ could still be badly biased in finite samples. We consider a second-order term in the Taylor series expansion to bias estimation that evaluates the nonlinearity of the ratio estimator θ and we propose a bias-corrected estimator, which is identical to the almost unbiased ratio estimator proposed by Tin (1965). Third, we investigate the problem of approximating the variance of a nonlinear function of parameters based on a second-degree Taylor series expansion. Unfortunately, when calculating the variance of the second-degree Taylor expansion, most authors (Parr, 1983; Hayya et al., 1975; Y. Wang et al., 2015) did not take into account the possible covariances between the random variables, which is indispensable because it provides a better approximation. This variance is none other than the variance of the bias-corrected estimator (or the variance of the almost unbiased ratio estimator of Tin (1965)). Fourth, we define a modified version of the Delta method, correct the estimator of the bias, and calculate the corresponding variance. This can be helpful in terms of more accurate coverage probabilities for the CIs. Fifth, we propose a novel analytical approach to construct the CI for the ratio estimate. Our method, Edgeworth expansion with bias-corrected estimator uses the Edgeworth expansion but adopts an estimator corrected for the bias and its variance. The method always produces a bounded CI. Simulation results show that it generally outperforms the Edgeworth expansion in terms of controlling the coverage probabilities and the average width and is particularly useful when the data are skewed.

The rest of this paper is organized as follows: Section 2 presents some highlights. Section 3 studies the different methods for constructing CIs, the Fieller and Delta methods

Econometrics **2025**, 13, 37 4 of 27

and we will develop the Edgeworth expansions for the Delta method. Section 4 provides an analytical form of the bias that can be used to construct the bias-corrected estimator and to calculate the variance of the bias-corrected estimator. Section 5 presents the confidence intervals with the bias-corrected estimator. Section 6 presents some econometric applications. The simulation study and the results are presented in Sections 7 and 8 conclude the paper.

2. Some Highlights

2.1. Definitions, Notation

Let X and Y be two random variables, we assume that the first and second moments exist, then the expected value of X is denoted by E(X) = x, the variance of X by V(X), the square of the coefficient of variation of X is defined by $CV(X)^2 = \frac{V(X)}{x^2}$ and the coefficient of variation of X by $CV(X) = \frac{\sqrt{V(X)}}{x}$. A similar notation will be used for the random variable Y. We assume that E(X) = x and E(Y) = y are non-zero. The covariance of X and Y is defined by Cov(X,Y) = E(XY) - E(X)E(Y), the correlation coefficient between X and Y is defined by $P = \frac{Cov(X,Y)}{\left(\sqrt{V(X)}\right)\left(\sqrt{V(Y)}\right)}$ so it satisfies $|P| \leq 1$ and $Cov(X,Y) = P\sqrt{V(X)}\sqrt{V(Y)}$.

The coefficient of co-variation of X and Y is defined by $CV(X,Y) = \frac{Cov(X,Y)}{xy}$, which can be expressed as the *product* of the correlation coefficient and the coefficients of variation of X and Y, respectively: $CV(X,Y) = \rho \frac{\sqrt{V(X)}}{x} \frac{\sqrt{V(Y)}}{y} = \rho CV(X)CV(Y)$. We use the notation $[a \pm b]$ for the interval [a - b, a + b] $(b \succ 0)$.

2.2. The Ratio Estimator Is Biased

Let $\widehat{\theta}_1$ and $\widehat{\theta}_2$ be consistent estimators of θ_1 and θ_2 , respectively, $E(\widehat{\theta}_1) = \theta_1$ and $E(\widehat{\theta}_2) = \theta_2$ and the ratio estimator $\widehat{\theta} = \widehat{\theta}_1/\widehat{\theta}_2$ is a consistent estimator of the ratio $\theta = \theta_1/\theta_2$. It is well known that the ratio of two unbiased estimators is not, in general, itself an unbiased estimator, i.e., $E(\widehat{\theta}_1/\widehat{\theta}_2) \neq E(\widehat{\theta}_1)/E(\widehat{\theta}_2) = \theta_1/\theta_2$.

The expected value of the ratio between $\hat{\theta}_1$ and $\hat{\theta}_2$, provided that the appropriate moments exist, is given by

$$E(\widehat{\theta}_1/\widehat{\theta}_2) = E(\widehat{\theta}_1 \times 1/\widehat{\theta}_2)$$

= $E(\widehat{\theta}_1) \times E(1/\widehat{\theta}_2) + Cov(\widehat{\theta}_1, 1/\widehat{\theta}_2)$

If $\hat{\theta}_1$ and $\hat{\theta}_2$ are independent or if $\hat{\theta}_1$ and $1/\hat{\theta}_2$ are uncorrelated, then

$$E(\widehat{\theta}_1 \times 1/\widehat{\theta}_2) = E(\widehat{\theta}_1) \times E(1/\widehat{\theta}_2).$$

It is well known that $E(1/\hat{\theta}_2) \neq 1/E(\hat{\theta}_2)$, Jensen's inequality implies that $E(1/\hat{\theta}_2) \geq 1/E(\hat{\theta}_2)$ because the function 1/z is convex for z > 0 or z < 0, then we have

$$E(\widehat{\theta}_1/\widehat{\theta}_2) = E(\widehat{\theta}_1) \times E(1/\widehat{\theta}_2) \ge E(\widehat{\theta}_1)/E(\widehat{\theta}_2)$$

and using that $E(\widehat{\theta}_1) = \theta_1$ and $E(\widehat{\theta}_2) = \theta_2$ we have

$$E(\widehat{\theta}_1/\widehat{\theta}_2) \ge \theta_1/\theta_2$$
$$E(\widehat{\theta}) \ge \theta$$

This result shows that the estimator of the ratio of two unbiased estimators is, in general, biased.

We will now consider a more general framework *that can be precisely define* the bias of the ratio estimator $\hat{\theta}$.

Econometrics **2025**, 13, 37 5 of 27

Note that the covariance of $\hat{\theta}_2$ and the ratio $\hat{\theta}_1/\hat{\theta}_2$ is

$$Cov(\widehat{\theta}_2, \widehat{\theta}_1/\widehat{\theta}_2) = E(\widehat{\theta}_2 \times \widehat{\theta}_1/\widehat{\theta}_2) - E(\widehat{\theta}_2) \times E(\widehat{\theta}_1/\widehat{\theta}_2)$$
$$= E(\widehat{\theta}_1) - E(\widehat{\theta}_2) \times E(\widehat{\theta}_1/\widehat{\theta}_2)$$

Then, by rearranging these terms, provided that $E(\widehat{\theta}_2) \neq 0$, we obtain the expected value of the ratio between $\widehat{\theta}_1$ and $\widehat{\theta}_2$

$$E(\widehat{\theta}_1/\widehat{\theta}_2) = E(\widehat{\theta}_1)/E(\widehat{\theta}_2) - 1/E(\widehat{\theta}_2) \times Cov(\widehat{\theta}_2, \widehat{\theta}_1/\widehat{\theta}_2)$$

This result shows that the expected value of a ratio of two random variables is not the ratio of the expected values and using that $E(\hat{\theta}_1) = \theta_1$ and $E(\hat{\theta}_2) = \theta_2$, we get

$$\begin{split} E(\widehat{\theta}_1/\widehat{\theta}_2) &= \theta_1/\theta_2 - 1/\theta_2 \times Cov(\widehat{\theta}_2,\widehat{\theta}_1/\widehat{\theta}_2) \\ E(\widehat{\theta}) &= \theta - 1/\theta_2 \times Cov(\widehat{\theta}_2,\widehat{\theta}) \end{split}$$

Then the bias of the ratio estimator $\hat{\theta}$ is

$$Bias(\widehat{\theta}) = E(\widehat{\theta}) - \theta = -1/\theta_2 \times Cov(\widehat{\theta}_2, \widehat{\theta})$$

The ratio estimator $\widehat{\theta}$ is thus generally a biased estimator of the true value of the ratio θ even if its components $\widehat{\theta}_1$ and $\widehat{\theta}_2$ are themselves unbiased with the size of the bias of $\widehat{\theta}$ depending on both θ_2 and the covariance between $\widehat{\theta}_2$ and the ratio $\widehat{\theta}_1/\widehat{\theta}_2$.

The bias of $\widehat{\theta}$ can be written as

$$\mathit{Bias}(\widehat{\theta}) = - \frac{\rho^* \sqrt{V(\widehat{\theta}_2)} \sqrt{V(\widehat{\theta})}}{\theta_2}$$

where $\rho^* = \frac{Cov(\widehat{\theta}_2,\widehat{\theta})}{\left(\sqrt{V(\widehat{\theta}_2)}\right)\left(\sqrt{V(\widehat{\theta})}\right)}$ is the correlation coefficient between $\widehat{\theta}_2$ and the ratio estimator $\widehat{\theta}$ and $\sqrt{V(\widehat{\theta}_2)}$ and $\sqrt{V(\widehat{\theta})}$ are their standard errors, respectively.

Consequently, the absolute value of the bias is

$$\left|\mathit{Bias}(\widehat{\theta})\right| = \frac{|-\rho^*|\sqrt{V(\widehat{\theta}_2)}\sqrt{V(\widehat{\theta})}}{\theta_2} \leq \frac{\sqrt{V(\widehat{\theta}_2)}\sqrt{V(\widehat{\theta})}}{\theta_2}$$

assuming $\theta_2 > 0$ and the correlation coefficient between $\widehat{\theta}_2$ and the ratio estimator $\widehat{\theta}$ satisfies $|\rho^*| \leq 1$.

Thus, an upper bound to the ratio of the absolute value of the bias to its standard error is given by

$$\left| \frac{Bias(\widehat{\theta})}{\sqrt{V(\widehat{\theta})}} \right| \le \frac{\sqrt{V(\widehat{\theta}_2)}}{\theta_2} = CV(\widehat{\theta}_2)$$

where $CV(\widehat{\theta}_2)$ is the coefficient of variation of $\widehat{\theta}_2$.

The bias in the ratio estimator $\widehat{\theta}$ is negligible in relation to its standard error if the coefficient of variation of $\widehat{\theta}_2$ is small, which is likely to be the case when the sample size is sufficiently large. It is well known that the variance of estimator $V(\widehat{\theta}_2)$ is order $O(n^{-1})$ then also the bias of $\widehat{\theta}$ is also order $O(n^{-1})$. Cochran (1977) has shown that if the coefficient of variation of $\widehat{\theta}_2$ is less than 0.1, then the bias in ratio estimator $\widehat{\theta}$ is small relative to its standard error. However, in econometrics and statistics models, in which the bias

Econometrics **2025**, 13, 37 6 of 27

may be considerable and bias correction can often improve the finite sample performace of estimators.

Furthermore, the main difficulty in estimating the bias of $\hat{\theta}$ in order to obtain unbiased estimates of θ is to estimate the covariance between $\hat{\theta}_2$ and the ratio $\hat{\theta}_1/\hat{\theta}_2$. Thus, it is difficult to obtain an analytical expression of the bias, as we will see later in Section 4 that using an approximation of the ratio of the parameters gives an analytical form of the bias and leads to a reduction in the bias from order $O(n^{-1})$ to order $O(n^{-2})$.

3. Methods

3.1. The Delta Method (or the Taylor's Series Expansion)

The Delta method often referred to as the Taylor's series expansion (Dorfman et al., 1990; Briggs & Fenn, 1998; Li & Maddala, 1999, among others) estimates the variance of a nonlinear function of two or more random variables is given by taking a first-order Taylor expansion around the mean value of the variables and calculating the variance for this expression. In the case of the ratio of *parameters* $\hat{\theta} = g(\hat{\theta}_1, \hat{\theta}_2) = \hat{\theta}_1/\hat{\theta}_2$, the variance of $\hat{\theta}$ is (Full derivation details can be see in Appendix A).

$$V(\widehat{\theta}) = \frac{1}{\widehat{\theta}_2^2} \left[V(\widehat{\theta}_1) - 2 \left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2} \right) Cov(\widehat{\theta}_1, \widehat{\theta}_2) + \left(\frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \right) V(\widehat{\theta}_2) \right],$$

which can also be written as

$$V(\widehat{\theta}) = \frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \left[CV(\widehat{\theta}_1)^2 - 2CV(\widehat{\theta}_1, \widehat{\theta}_2) + CV(\widehat{\theta}_2)^2 \right]$$

where $CV(\widehat{\theta}_i)^2$ is the square of the coefficient of variation for a random variable $\widehat{\theta}_i$ for i=1,2 and $CV(\widehat{\theta}_1,\widehat{\theta}_2)=\rho CV(\widehat{\theta}_1)CV(\widehat{\theta}_2)$ is the coefficient of co-variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ and $\rho=\frac{Cov(\widehat{\theta}_1,\widehat{\theta}_2)}{\left(\sqrt{V(\widehat{\theta}_1)}\right)\left(\sqrt{V(\widehat{\theta}_2)}\right)}$ is the correlation coefficient between $\widehat{\theta}_1$ and $\widehat{\theta}_2$.

To construct a confidence interval for the ratio $\theta = \theta_1/\theta_2$, we assume that $n^{1/2}(\widehat{\theta} - \theta)$ is asymptotically normal distributed with zero mean and variance $V(\widehat{\theta})$.

Let $\widehat{V}(\widehat{\theta})$ be a consistent estimator of $V(\widehat{\theta})$, the Delta method $100(1-\alpha)\%$ confidence limits for the ratio θ_1/θ_2 is given as follows:

$$CI_D: \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \pm z_{\alpha/2}Q_D$$

where $\frac{\widehat{\theta}_1}{\widehat{\theta}_2}$ is namely the classical estimator and $Q_D = \sqrt{\widehat{V}(\widehat{\theta})} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2}$ $\left[\widehat{CV}(\widehat{\theta}_1)^2 - 2\widehat{CV}(\widehat{\theta}_1,\widehat{\theta}_2) + \widehat{CV}(\widehat{\theta}_2)^2\right]^{1/2}$, the estimated standard error of the classical estimator and $z_{\alpha/2}$ is the $(\alpha/2)$ th quantile for standard normal distribution.

This method assumes that $\widehat{\theta}$ is normally distributed and thus symmetrical around its mean. However, the assumption of normality is clearly strong as there is no guarantee that $\widehat{\theta}$ is normally distributed.

However, for large sample sizes (or rather small coefficients of variation) the distribution of a ratio may be close to normal.

The assumption of a normal distribution may be justified in the case of large samples, but it is unlikely that the distribution of a ratio will generally follow a well-behaved distribution. Furthermore, the assumption of a normal distribution may be quite inaccurate if the data have a skewed distribution.

Econometrics **2025**, 13, 37 7 of 27

3.2. The Fieller Method

Fieller (1954) proposed a general procedure for constructing confidence limits for the ratio of the means of two normal distributions. In Fieller's method, the ratio variable is transformed into a linear function. The confidence interval of the ratio variable can be obtained by solving out the quadratic roots of the linear function.

For testing the null hypothesis $H_0: \frac{\theta_1}{\theta_2} = \gamma$ equivalently it is written as on a linear combination of the parameters $H_0': \theta_1 - \gamma \theta_2 = 0$, the method assumes that $\widehat{\theta}_1$ and $\widehat{\theta}_2$ follow a joint normal distribution function such that $\widehat{\theta}_1 - \gamma \widehat{\theta}_2$ is normally distributed. Hence, the pivotal statistic for this test is

$$T = \frac{\widehat{\theta}_1 - \gamma \widehat{\theta}_2}{\sqrt{\widehat{V}(\widehat{\theta}_1) - 2\gamma \widehat{Cov}(\widehat{\theta}_1\widehat{\theta}_2) + \gamma^2 \widehat{V}(\widehat{\theta}_2)}}$$

which is *t*-distribution with *df* degrees of freedom under the null hypothesis.

Let $t_{\alpha/2,df}$ denotes the $100(1-\alpha/2)$ th percentile of the t- distribution with df degrees of freedom, we have

$$P\left[T^2 \le t_{\alpha/2,df}^2\right] = 1 - \alpha$$

By replacing the expression of square T and rearranging gives a quadratic equation in γ .

$$a\gamma^2 + b\gamma + c \leq 0$$

where
$$a = \widehat{\theta}_2^2 - t_{\alpha/2,df}^2 \widehat{V}(\widehat{\theta}_2)$$
, $b = -2 \Big(\widehat{\theta}_1 \widehat{\theta}_2 - t_{\alpha/2,df}^2 \widehat{Cov}(\widehat{\theta}_1 \widehat{\theta}_2) \Big)$, and $c = \Big(\widehat{\theta}_1^2 - t_{\alpha/2,df}^2 \widehat{V}(\widehat{\theta}_1) \Big)$. Finding an explicit form for the confidence intervals for γ requires solving the

Finding an explicit form for the confidence intervals for γ requires solving the quadratic equation. The solution of this inequality depends on the sign of a and $d = b^2 - 4ac$, the discriminant of the quadratic equation. Through simple calculation, we can expressed d as follows:

$$d = 4\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)^2 t_{\alpha/2,df}^2 \left\{ \left[\widehat{CV}(\widehat{\theta}_2) - \widehat{\rho}\widehat{CV}(\widehat{\theta}_1)\right]^2 + a\widehat{CV}(\widehat{\theta}_1)^2 (1 - \widehat{\rho}^2) \right\}$$

where $\hat{\rho}$ is the estimate of the correlation coefficient between $\hat{\theta}_1$ and $\hat{\theta}_2$. Hence, a > 0 also implies d > 0.

If d>0, let γ_L and γ_U ($\gamma_L<\gamma_U$) be the two real-valued solutions to the quadratic equation in γ by changing the inequality into an equality. This gives the bounds of the Fieller interval in the case a>0. These two roots are the lower and upper limits of the $(1-\alpha)$ confidence interval. The bounds of the interval are given by

$$CI_F: [\gamma_L, \gamma_U] = \left(\frac{\widehat{ heta}_1}{\widehat{ heta}_2}\right)_F \pm t_{lpha/2, df} Q_F$$

where $\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_F = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left[\frac{1}{1-h} \left(1 - h\widehat{\rho} \frac{\widehat{CV}(\widehat{\theta}_1)}{\widehat{CV}(\widehat{\theta}_2)}\right)\right]$ is the Fieller estimator and $Q_F = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \frac{1}{1-h} \left[\widehat{CV}(\widehat{\theta}_1)^2 - 2\widehat{CV}(\widehat{\theta}_1, \widehat{\theta}_2) + \widehat{CV}(\widehat{\theta}_2)^2 - h\widehat{CV}(\widehat{\theta}_1)^2 (1 - \widehat{\rho}^2)\right]^{1/2}$ the estimated standard error of the Fieller estimator and $h = t_{\alpha/2, df}^2 \widehat{CV}(\widehat{\theta}_2)^2$.

However, if a < 0 the Fieller CI will be unbounded. Hence, if d > 0 the Fieller CI will be the complement of a finite interval $(-\infty, \gamma_U) \cup (\gamma_L, \infty)$ and if d < 0 the Fieller CI will be the whole real line $(-\infty, +\infty)$.

Other intervals may be considered when a=0, the Fieller CI will be $\left]-\infty, \frac{-c}{b}\right]$ if b>0 otherwise, it will be $\left[\frac{-c}{b}, \infty\right[$ if b<0.

Econometrics **2025**, 13, 37 8 of 27

Remark 1.

(1) Following the Fieller estimator, the term $\frac{1}{1-h}\left(1-h\widehat{\rho}\frac{\widehat{CV}(\widehat{\theta}_1)}{\widehat{CV}(\widehat{\theta}_2)}\right)$ can be considered as a correction factor to the estimated ratio estimator.

- (2) If $\widehat{\rho}_{\widehat{CV}(\widehat{\theta}_2)}^{\widehat{CV}(\widehat{\theta}_1)} = 1$ then the Fieller estimator $\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_F$ is equal to the classical estimator $\frac{\widehat{\theta}_1}{\widehat{\theta}_2}$.
- (3) In the case of finite interval, the condition a > 0 is equivalent to $\left| \frac{\widehat{\theta}_2}{\sqrt{\widehat{V}(\widehat{\theta}_2)}} \right| > t_{\alpha/2,df}$ which means rejecting the null hypothesis $H_0: \theta_2 = 0$, i.e., θ_2 is significantly different from zero. The test of this null hypothesis is the first step of Scheffé's procedure (Scheffé, 1970).
- (4) The t-statistic $\left|\frac{\widehat{\theta}_2}{\sqrt{\widehat{V}(\widehat{\theta}_2)}}\right|$ is equal to $\left|\frac{1}{\widehat{CV}(\widehat{\theta}_2)}\right|$ the absolute inverse of the coefficient of variation for $\widehat{\theta}_2$, so the null hypothesis is rejected if the coefficient of variation for $\widehat{\theta}_2$ is negligible. (A high coefficient of variation for $\widehat{\theta}_2$ means a low statistical value).
- (5) If h is close to zero the Fieller CI becomes the Delta CI.

It should be noted that the null hypothesis H_0' : $\theta_1 - \gamma \theta_2 = 0$, was obtained from the nonlinear relationship $\frac{\theta_1}{\theta_2} = \gamma$ only when $\theta_2 \neq 0$. However, Fieller's method does not take this information into account. Therefore, the Fieller CI has the potential to overestimate the confidence length.

Furthermore, Fieller's estimator is a linear combination of the ratio estimator (or classical estimator). As we mentioned in Section 1, the ratio estimator is generally biased, so Fieller's estimator is also generally biased.

The advantage of Fieller's method over the Delta method is that it takes into account the potential skewness of the sampling distribution of the ratio estimator and therefore may not be symmetric around the point estimate. Fieller's method provides an exact solution subject to the joint normality assumption. However, it has been argued that the assumption of joint normality may be difficult to justify, particularly when sample sizes are small. In particular, the random variable follows a skewed distribution, which may cause problems for the normality assumption.

The normal approximation is a rather rough approximation, especially when sample sizes are not large; it does not take into account the skewness of the underlying distribution, which is often the main source of error of the normal approximation. To remove the effect of the skewness, we develop the Edgeworth expansion.

3.3. Edgeworth Expansion

The Delta method-based confidence interval is not very robust and can be quite inaccurate in practice for non-normal data. It produces intervals that are symmetric around the point estimate, so it does not take skewness into account. The correction for skewness used in our confidence intervals is based on the Edgeworth expansion.

We propose a method based on the Edgeworth expansion to modify the Delta intervals to remove the effect of skewness. The expansion provides a way to correct for the skewness in the data and to derive new confidence intervals for the ratio parameters. Thus, we consider two aspects: first an Edgeworth expansion is derived for the Delta method for a ratio of parameters on a normal random variable and second by using the inverse of the Edgeworth expansions, which are the quantiles of the distribution that is the Cornish–Fisher expansion, we construct an approximate confidence interval, which contains a $n^{-1/2}$ order correction for the effect of skewness.

The Delta method can be easily extended for a better approximation by using Edgeworth expansion.

Econometrics **2025**, 13, 37 9 of 27

Let $U = \widehat{V}(\widehat{\theta})^{-1/2} \sqrt{n}(\widehat{\theta} - \theta)$ where $\widehat{\theta} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2}$, $\theta = \frac{\theta_1}{\theta_2}$ and $\widehat{V}(\widehat{\theta})$ the estimate of $V(\widehat{\theta})$ in Delta method, we assume that the distribution of a random variable U has the Edgeworth expansion (Hwang, 2019; Hall, 1992a).

$$F(x) = P(U \le x) = \Phi(x) - n^{-1/2} \kappa \frac{1}{6} (x^2 - 1) \phi(x) + O(n^{-1/2})$$

where $\Phi(x)$ and $\phi(x)$ are the standard normal distribution and density functions, respectively, κ is the skewness, and n is the sample size. This expansion can be interpreted as the sum of the normal distribution $\Phi(x)$, and an error due to the skewness of the distribution. When the error (the $n^{-1/2}$ skewness correction) in absolute value is small, U can be accurately approximated by a normal distribution. Conversely, when the error in absolute value is large, the second term in the formulation cannot be ignored, and therefore, the normal approximation would not be as accurate. The $n^{-1/2}$ skewness correction is an even function of x, which means that it changes the distribution function symmetrically about zero. Thus, the skewness of the distribution F has a significant effect, especially when the sample size n is small.

To construct asymptotic confidence intervals, we should invert the Edgeworth expansions to obtain expansions of distribution quantiles. Such expansions are known as Cornish–Fisher expansions.

For any $0 < \alpha < 1$, let ξ_{α} be the $\alpha - th$ quantile of distribution F(.), which is the solution to $F(\xi_{\alpha}) = \alpha$. This quantile of distribution $\xi_{\alpha} = F^{-1}(\alpha)$ admits a Cornish–Fisher expansion of the form (Hall, 1992b; Hwang, 2019).

$$\xi_{\alpha} = z_{\alpha} + n^{-1/2} \hat{\kappa} \frac{1}{6} (z_{\alpha}^2 - 1) + O(n^{-1/2})$$

where $\hat{\kappa}$ is the estimate of κ and z_{α} is the α -th quantile of the standard normal distribution. The $100(1-\alpha)\%$ Edgeworth expansion confidence interval for the ratio $\frac{\theta_1}{\theta_2}$ is given by

$$CI_E: \left[\frac{\widehat{\theta}_1}{\widehat{\theta}_2} - \xi_{1-\alpha/2} Q_D, \quad \frac{\widehat{\theta}_1}{\widehat{\theta}_2} - \xi_{\alpha/2} Q_D \right]$$

where $Q_D = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \Big[\widehat{CV}(\widehat{\theta}_1)^2 - 2\widehat{CV}(\widehat{\theta}_1, \widehat{\theta}_2) + \widehat{CV}(\widehat{\theta}_2)^2 \Big]^{1/2}$, and $\xi_{\alpha/2}$ and $\xi_{1-\alpha/2}$ are the $(\alpha/2)th$ and $(1-\alpha/2)th$ quantiles of distribution F(.).

For positively skewed data, the true $1 - \alpha/2$ quantile $\xi_{1-\alpha/2}$ is larger than the associated standard normal quantiles $z_{\alpha/2}$ and similarly the true lower quantile $\xi_{\alpha/2}$ is larger than $-z_{\alpha/2}$.

From the Cornish–Fisher expansion, we can state the asymptotic coverage probability of the proposed intervals.

The coverage probability of confidence intervals is given by

$$P(\frac{\theta_1}{\theta_2} \in CI_E) = 1 - \alpha + O(n^{-1/2}).$$

4. Bias-Correction Analysis

4.1. Bias of Estimator

In Section 1, we showed that the ratio estimator $\hat{\theta} = \hat{\theta}_1 / \hat{\theta}_2$ is a biased estimator of the ratio parameters. It is essential to determine the expected direction and magnitude of this bias.

The Fieller estimator and the classical estimator are strongly consistent (converge to the ratio θ_1/θ_2 with probability one), and generally are biased.

In the following, we propose to correct the bias of the classic estimator. It is well known in the literature that the ratio of the parameters uses only first-order expansions to approximate asymptotic sampling distributions. However, calculating higher-order expansions can also be useful given that they can be used to estimate the bias of the ratio of the parameters and the analytical form of the bias obtained can be used to construct the bias-corrected estimator. (Furthermore, higher-order expansions are also useful because they can be used to estimate the bias of the ratio parameters and the analytical form of the bias obtained can be used to construct the bias-corrected estimator.)

We consider a second-order term in the Taylor series expansion to bias estimation that evaluates the nonlinearity of the ratio of the parameters. This additional second-order term can be helpful, in the sense of more accurate coverage probabilities for the CIs.

Let θ is $g(\theta_1, \theta_2) = \theta_1/\theta_2$ then from a second-order Taylor's series expansion

$$g(\widehat{\theta}_1, \widehat{\theta}_2) = g(\theta_1, \theta_2) + G'\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix} + \frac{1}{2}\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix}' H\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix} + R_n$$

where *G* is a Jacobian vector containing all the first-order partial derivatives and *H* is a Hessian matrix containing all the second partial derivatives for the nonlinear function

$$g(\widehat{\theta}_1, \widehat{\theta}_2)$$
 evaluated at θ_1 and θ_2 , and the remainder R_n is of order $O\left(\left\|\begin{array}{c} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{array}\right\|^2\right)$ i.e., $\frac{R_n}{\left\|\begin{array}{c} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{array}\right\|^2} \longrightarrow 0$ as $\widehat{\theta}_i \longrightarrow \theta_i$ for $i = 1, 2$.as $n \longrightarrow \infty$.

We define the bias and variance of ratio estimator using the first and second moments of the terms in this second-order Taylor's series expansion. Taking the expectation of this expansion and under the conditions $E(\widehat{\theta}_i - \theta_i) = 0$ for i = 1, 2 we obtain the bias of ratio estimator given in the following proposition.

Proposition 1. Let $\theta = \frac{\theta_1}{\theta_2}$ be a ratio of parameters, a second-order Taylor's series expansion gives the approximation of bias

$$Bias(\widehat{\theta}) = E(\widehat{\theta}) - \theta = \frac{1}{2}(vecH)'vec(\Sigma) + O(n^{-2})$$

where Σ is the variance-covariance matrix of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ and H is the Hessian matrix of the second partial derivatives.

The estimate of bias is given by

$$\widehat{Bias}(\widehat{\theta}) = \frac{1}{2} (vec\widehat{H})' vec\widehat{\Sigma} + O(n^{-2})$$

where \widehat{H} and $\widehat{\Sigma}$ are the estimates of H and Σ respectively.

This yields

$$\widehat{Bias}(\widehat{\theta}) = -\frac{1}{\widehat{\theta}_2^2} \widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2) + \frac{\widehat{\theta}_1}{\widehat{\theta}_2^3} \widehat{V}(\widehat{\theta}_2) + O(n^{-2}),$$

which can also be written as

$$\widehat{Bias}(\widehat{\theta}) = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left[\frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2} - \frac{\widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2)}{\widehat{\theta}_2 \widehat{\theta}_1} \right] + O(n^{-2})$$

where $\frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2} - \frac{\widehat{Cov}(\widehat{\theta}_1,\widehat{\theta}_2)}{\widehat{\theta}_2\widehat{\theta}_1}$ can be considered as a correction factor to the estimated ratio estimator.

Proof. (see Appendix A). \Box

This bias is identical to Tin's bias, as described by Tin in 1965. It uses the same information as the correction factor formed by subtracting $\frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2}$ from $\frac{\widehat{Cov}(\widehat{\theta}_2,\widehat{\theta}_1)}{\widehat{\theta}_2\widehat{\theta}_1}$. This bias is order to $O(n^{-2})$ Tin (1965). Our bias is derived by a different method. Tin (1965) and David and Sukhatme (1974) used an asymptotic series expansion of the ratio estimator under certain conditions. The high-order of Tin's bias formulation was given by David and Sukhatme (1974).

To obtain the sign of the bias, we express the bias as a function of the coefficient of variation and the coefficient of co-variation

$$\widehat{Bias}(\widehat{\theta}) = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left(\widehat{CV}(\widehat{\theta}_1) \widehat{CV}(\widehat{\theta}_2) \left[\frac{\widehat{CV}((\widehat{\theta}_2))}{\widehat{CV}(\widehat{\theta}_1)} - \widehat{\rho} \right] \right) + O(n^{-2})$$

where $\hat{\rho}$ is the estimate of the correlation coefficient between $\hat{\theta}_1$ and $\hat{\theta}_2$.

Following this latter formula, if the coefficient of variation of $\widehat{\theta}_2$ is close to zero, then the bias may be negligible relative to the variation in $\widehat{\theta}$. Furthermore, if the coefficient of variation of $\widehat{\theta}_2$ is greater than the coefficient of variation of $\widehat{\theta}_1$ the absolute value of the bias increases if the correlation between $\widehat{\theta}_1$ and $\widehat{\theta}_2$ becomes zero or negative. Similarly, if the coefficient of variation of $\widehat{\theta}_1$ is greater than the coefficient of variation of $\widehat{\theta}_2$ the bias is negative for a high positive correlation coefficient. Moreover, if $\frac{\widehat{CV}((\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)} > \widehat{\rho}$ then the absolute value of the bias is positive, if $\frac{\widehat{CV}((\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)} < \widehat{\rho}$ then the absolute value of the bias is negative, and if $\frac{\widehat{CV}((\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)} = \widehat{\rho}$ then the ratio estimator is unbiased.

4.2. The Bias-Corrected Estimator

We have obtained an analytic form of the bias and the estimate bias of the ratio parameters can be used to correct the estimator, the bias-corrected estimator is given by

$$\widehat{\theta}_{BC} = \widehat{\theta} - \widehat{Bias}(\widehat{\theta}) = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} - \frac{1}{2}(vec\widehat{H})'vec\widehat{\Sigma} + O(n^{-2})$$

This yields

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} + \frac{1}{\widehat{\theta}_2^2} \widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2) - \frac{\widehat{\theta}_1}{\widehat{\theta}_2^3} \widehat{V}(\widehat{\theta}_2) + O(n^{-2})$$

Following this result, the bias-corrected estimator is obtained by adjusting the classical estimator by the term that is capable of reducing it from order $O(n^{-1})$ to order $O(n^{-2})$.

The bias-corrected estimator can also be written as

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left\{ 1 + \left[\frac{\widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2)}{\widehat{\theta}_1 \widehat{\theta}_2} - \frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2} \right] \right\} + O(n^{-2})$$

where $1 + \left[\frac{\widehat{Cov}(\widehat{\theta}_1,\widehat{\theta}_2)}{\widehat{\theta}_1\widehat{\theta}_2} - \frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2}\right]$ can be considered as a correction factor to the estimated ratio estimator.

This bias-corrected estimator $\widehat{\theta}_{BC}$ has the same structure as Tin's (1965) almost unbiased ratio estimator in the sense that its bias is of $O(n^{-2})$, i.e., the bias of $\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC}$ converges to zero at a fast rate than that of $\frac{\widehat{\theta}_1}{\widehat{\theta}_2}$. Tin called it a modified ratio estimator". He showed that his estimator is better than other competing estimators of population mean, up to

the second order of approximation and it is equivalent to the Beale (1962) estimator up to the first order of approximation. Tin's estimator has been studied theoretically and via simulation by, Dalabehera and Sahoo (1995), Swain and Dash (2020) and they found Tin's estimator generally to be less biased and more efficient compared with other proposed ratio estimators.

The bias-corrected estimator $\widehat{\theta}_{BC}$ in terms of coefficient of variation and the coefficient of co-variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ is

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left\{ 1 + \widehat{CV}(\widehat{\theta}_1) \widehat{CV}(\widehat{\theta}_2) \left[\widehat{\rho} - \frac{\widehat{CV}(\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)} \right] \right\} + O(n^{-2})$$

where $1 + \widehat{CV}(\widehat{\theta}_1)\widehat{CV}(\widehat{\theta}_2)\left[\widehat{\rho} - \frac{\widehat{CV}(\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)}\right]$ can be considered as a correction factor to the estimated ratio estimator.

In the next, we examine the case where the numerator and denominator of a ratio are independent. In this case, we will specify the bias and the bias-corrected estimator in the following proposition:

Proposition 2. If $\hat{\theta}_1$ and $\hat{\theta}_2$ are independent, we have the following

(i) The estimate of the bias is $\widehat{Bias}(\widehat{\theta}) = \frac{\widehat{\theta}_1}{\widehat{\theta}_2}\widehat{CV}(\widehat{\theta}_2)^2 = \frac{\widehat{\theta}_1}{\widehat{\theta}_2}\frac{1}{t(\widehat{\theta}_2)^2}$

where $t(\widehat{\theta}_2)^2$ denotes the square of the t-statistic (or F_1 statistic) for $\widehat{\theta}_2$ and $\frac{1}{t(\widehat{\theta}_2)^2}$ can be considered as a correction factor to the estimated ratio estimator.

The estimate of the bias of the ratio parameters is an estimator of the ratio weighted by the square of the coefficient of variation of $\hat{\theta}_2$ (the inverse of the square of the t-statistic for $\hat{\theta}_2$ or the inverse of the F_1 statistic).

(ii) The bias-corrected estimator is
$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left[1 - \widehat{CV}(\widehat{\theta}_2)^2\right] = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left[1 - \frac{1}{t(\widehat{\theta}_2)^2}\right]$$
.

The bias-corrected estimator of the ratio parameters is an estimator of the ratio weighted by the simple statistic $\left[1-\frac{1}{t(\widehat{\theta}_2)^2}\right]$, this weight is less than one because $\widehat{CV}(\widehat{\theta}_2)^2$ or $\frac{1}{t(\widehat{\theta}_2)^2}$ is positive.

4.3. The Variance of the Bias-Corrected Estimator

As we have shown, the bias-corrected estimator $\widehat{\theta}_{BC}$ corresponds to the Tin (1965) almost unbiased ratio estimator, also known as the modified ratio estimator. The approximation of the variance of $\widehat{\theta}$ with a second-order term expressed in terms of the coefficient of variation and the coefficient of co-variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ is identical to the variance of the almost unbiased ratio estimator. We therefore use this variance as the variance of the bias-corrected estimator.

The estimate of the variance of the bias-corrected estimator $\widehat{\theta}_{BC}$ is as follows (full derivation details can be see in Appendix A):

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \underbrace{\widehat{G}'\widehat{\Sigma}\widehat{G}}_{\text{first-order part}} + \underbrace{\frac{1}{2}(vec\widehat{H})'(\widehat{\Sigma}\otimes\widehat{\Sigma})vec\widehat{H}}_{\text{second-oder part}}$$

where the first order part $\hat{G}'\hat{\Sigma}\hat{G}$ is the estimate of the variance of $\hat{\theta}$ corresponding to a first order approximation and the second order part corresponding to an additional part from second-order approximation, which permit to take into account the correlation between the random variables.

This yields

$$\begin{split} \widehat{V}\Big[\big(\widehat{\theta}_{BC}\big)\Big] &= \underbrace{\frac{1}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \left[\widehat{V}\big(\widehat{\theta}_{1}\big) - 2\bigg(\frac{\widehat{\theta}_{1}}{\widehat{\theta}_{2}}\bigg)\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2}) + \bigg(\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}\bigg)\widehat{V}(\widehat{\theta}_{2})\right]_{\text{first-order approximation}} \\ &+ \underbrace{\frac{1}{\widehat{\theta}_{2}^{4}}\widehat{V}(\widehat{\theta}_{2})}_{\text{first-order approximation}} \left[\widehat{V}(\widehat{\theta}_{1}) - 4\bigg(\frac{\widehat{\theta}_{1}}{\widehat{\theta}_{2}}\bigg)\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2}) + 2\bigg(\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}\bigg)\widehat{V}(\widehat{\theta}_{2})\right] + \frac{1}{\widehat{\theta}_{2}^{4}}\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})^{2} \end{split}$$

additional part from second-order approximation

which can also be written as

$$\begin{split} \widehat{V}\left[(\widehat{\theta}_{BC})\right] &= \frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}} \left\{ \begin{array}{c} \underbrace{\left[\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} - 2\frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})}{\widehat{\theta}_{1}\widehat{\theta}_{2}} + \frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}\right]}_{\text{first-order approximation}} \\ &+ \underbrace{\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}} \left[\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} - 4\frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})}{\widehat{\theta}_{1}\widehat{\theta}_{2}} + 2\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}\right] + \frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})^{2}}{\widehat{\theta}_{1}^{2}\widehat{\theta}_{2}^{2}}} \\ &= \text{additional part from second-order approximation} \end{split} \right\}$$

Thus, this variance can be express in terms of coefficient variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ by

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}} \left\{ \begin{array}{c} \underbrace{\left[\widehat{CV}(\widehat{\theta}_{1})^{2} - 2\widehat{\rho}\widehat{CV}(\widehat{\theta}_{1})\widehat{CV}(\widehat{\theta}_{2}) + \widehat{CV}(\widehat{\theta}_{2})^{2}\right]}_{\text{first-order approximation}} \\ + \underbrace{\widehat{CV}(\widehat{\theta}_{2})^{2}\left[\widehat{CV}(\widehat{\theta}_{1})^{2} - 4\widehat{\rho}\widehat{CV}(\widehat{\theta}_{1})\widehat{CV}(\widehat{\theta}_{2}) + \widehat{\rho}^{2}\widehat{CV}(\widehat{\theta}_{1})^{2} + 2\widehat{CV}(\widehat{\theta}_{2})^{2}\right]}_{\text{additional part from second-order approximation}} \right\}$$

where $\hat{\rho}$ is the estimate of the correlation coefficient between $\hat{\theta}_1$ and $\hat{\theta}_2$.

This variance is identical to the variance of the "almost unbiased ratio estimator" (or the variance of the modified ratio estimator) of Tin (1965), see also David and Sukhatme (1974). If $\hat{\theta}_1$ and $\hat{\theta}_2$ are independent, we have:

(i) The estimate of the variance of the bias-corrected estimator $\hat{\theta}_{BC}$ is given by

$$\widehat{V}\Big[\big(\widehat{\theta}_{BC}\big)\Big] = \underbrace{\frac{1}{\widehat{\theta}_2^2}\left[\widehat{V}(\widehat{\theta}_1) + \left(\frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2}\right)\widehat{V}(\widehat{\theta}_2)\right]}_{\text{first-order approximation}} + \underbrace{\frac{1}{\widehat{\theta}_2^4}\widehat{V}(\widehat{\theta}_2)\left[\widehat{V}(\widehat{\theta}_1) + 2\left(\frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2}\right)\widehat{V}(\widehat{\theta}_2)\right]}_{\text{additional part from second-order approximation}}$$

which can also be written as

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}} \left\{ \underbrace{\left[\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} + \frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}\right]}_{\text{first-order approximation}} + \underbrace{\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}\left[\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} + 2\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}\right]}_{\text{additional part from second-order approximation}} \right\}$$

(ii) The variance $\widehat{V}\left[(\widehat{\theta}_{BC})\right]$ can be express in terms of coefficient variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}} \left\{ \underbrace{\left[\widehat{CV}(\widehat{\theta}_{1})^{2} + \widehat{CV}(\widehat{\theta}_{2})^{2}\right]}_{\text{first-order approximation}} + \underbrace{\widehat{CV}(\widehat{\theta}_{2})^{2}\left[\widehat{CV}(\widehat{\theta}_{1})^{2} + 2\widehat{CV}(\widehat{\theta}_{2})^{2}\right]}_{\text{additional part from second-order approximation}} \right\}$$

5. Confidence Intervals with Bias-Corrected Estimator

In this section, we construct new confidence intervals that take into account the bias of the estimator for the Delta method, and both the bias of the estimator and the asymmetry of the distribution for the Edgeworth expansion method.

5.1. Delta Method Based Confidence Interval with Bias-Corrected Estimator

Let us define the estimated standard error of the bias-corrected estimator $\hat{\theta}_{BC}$ by

$$Q_{BC} = \sqrt{\widehat{V}\left[(\widehat{\theta}_{BC})\right]}$$

And the bias-corrrected estimator is

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left\{ 1 + \left[\frac{\widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2)}{\widehat{\theta}_1 \widehat{\theta}_2} - \frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2} \right] \right\}$$

or in terms of coeffficient of variation and coeffficient of co-variation

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left\{ 1 + \widehat{CV}(\widehat{\theta}_1) \widehat{CV}(\widehat{\theta}_2) \left[\widehat{\rho} - \frac{\widehat{CV}(\widehat{\theta}_2)}{\widehat{CV}(\widehat{\theta}_1)} \right] \right\}.$$

The $100(1 - \alpha)\%$ confidence limits of the Delta method bias-corrected for the ratio θ_1/θ_2 is given as follows:

$$CI_{Dbc}: \left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} \pm z_{\alpha/2}Q_{BC}$$

where $z_{\alpha/2}$ is the $(\alpha/2)$ th quantile for standard normal distribution.

5.2. Edgeworth Expansion Based Confidence Interval with Bias-Corrected Estimator

For the Edgeworth expansion based confidence interval, we use the same correct term for the estimator of the ratio parameters, then the $100(1-\alpha)\%$ confidence interval for the ratio θ_1/θ_2 based Edgeworth expansion becomes

$$CI_{Ebc}: \left[\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2} \right)_{BC} - \xi_{1-\alpha/2} Q_{BC}, \ \left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2} \right)_{BC} - \xi_{\alpha/2} Q_{BC} \right]$$

where $\xi_{\alpha/2}$ and $\xi_{1-\alpha/2}$ are the $(\alpha/2)th$ and $(1-\alpha/2)th$ quantiles of distribution with

$$\xi_{\alpha} = z_{\alpha} + n^{-1/2} \hat{\kappa} \frac{1}{6} (z_{\alpha}^2 - 1)$$

where $\hat{\kappa}$ is the estimate of the skewness κ and z_{α} is the $\alpha - th$ quantile of the standard normal distribution.

6. Some Econometric Applications

6.1. The Ratio of Two Linear Combinations of Parameters

Many of the nonlinear functions studied in economic applications are expressed in the functional form of a ratio of two linear combinations of parameters. In this section, we consider the test of one such nonlinear function.

We will specify the bias of the estimator, the bias-corrected estimator, and its variance. Note that the formulations of the confidence intervals are given in the previous section. We will see that the calculations are quite simple and do not require intensive computation.

Econometrics **2025**, 13, 37 15 of 27

Consider the general linear model

$$Y = X\beta + \varepsilon$$

where Y is an $n \times 1$ vector of observations, X is a $n \times k$ full-rank design matrix, β is a $k \times 1$ vector of unknown parameters, and ε is an $n \times 1$ vector of normal random errors with zero mean and variance $\sigma^2 I$: $\varepsilon \sim N(0, \sigma^2 I)$. The OLS estimators of unknown parameters are $\widehat{\beta} = (X'X)^{-1}X'Y$ and $\widehat{\sigma}^2 = \widehat{\varepsilon}'\widehat{\varepsilon}/n - k$ where $\widehat{\varepsilon}$ are the OLS residuals.

Consider a null hypothesis for the ratio of two linear combinations of parameters

$$H_0 \ \theta = \frac{K'\beta}{L'\beta}$$

where *K* and *L* are $k \times 1$ vectors of known constants.

We have the following different terms:

$$\begin{array}{ll} \theta_1=K'\beta,\; \theta_1^2=(K'\beta)^2, & \widehat{V}(\widehat{\theta}_1)=K'\widehat{V}(\widehat{\beta})K=\widehat{\sigma}^2K'(X'X)^{-1}K\\ \theta_2=L'\beta,\; \theta_2^2=(L'\beta)^2,\; \theta_2^3=(L'\beta)^3,\; \widehat{V}(\widehat{\theta}_2)=L'\widehat{V}(\widehat{\beta})L=\widehat{\sigma}^2L'(X'X)^{-1}L\\ \theta_1\theta_2=(K'\beta)(L'\beta),\; \theta_1^2\theta_2^2=(K'\beta)^2(L'\beta)^2,\; \widehat{Cov}(\widehat{\theta}_1,\widehat{\theta}_2)=\widehat{Cov}(K'\widehat{\beta},L'\widehat{\beta})=\widehat{\sigma}^2K'(X'X)^{-1}L \end{array}$$

By replacing all these terms in the formulation of the bias for $\widehat{\theta}$, the bias-corrected estimator $\widehat{\theta}_{BC}$, and the variance of the bias-corrected estimator $\widehat{V}(\widehat{\theta}_{BC})$, we have the following proposition.

Proposition 3.

(i) The bias for $\widehat{\theta}$ is

$$\widehat{\mathit{Bias}}(\widehat{\theta}) = -\frac{1}{(L'\beta)^2}\widehat{\sigma}^2 K'(X'X)^{-1}L + \frac{K'\widehat{\beta}}{(L'\widehat{\beta})^3}\widehat{\sigma}^2 L'(X'X)^{-1}L,$$

which can also be written as

$$\widehat{Bias}(\widehat{\theta}) = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} \left[\frac{\widehat{\sigma}^2 L'(X'X)^{-1} L}{(L'\widehat{\beta})^2} - \frac{\widehat{\sigma}^2 K'(X'X)^{-1} L}{(K'\widehat{\beta})(L'\widehat{\beta})} \right]$$

where $\left[\frac{\widehat{\sigma}^2 L'(X'X)^{-1}L}{(L'\widehat{\beta})^2} - \frac{\widehat{\sigma}^2 K'(X'X)^{-1}L}{(K'\widehat{\beta})(L'\widehat{\beta})}\right]$ can be considered as a correction factor to the estimated

ratio estimator.

(ii) The bias-corrected estimator $\hat{\theta}_{BC}$ is given by

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} + \frac{1}{(L'\beta)^2}\widehat{\sigma}^2 K'(X'X)^{-1}L - \frac{K'\widehat{\beta}}{(L'\widehat{\beta})^3}\widehat{\sigma}^2 L'(X'X)^{-1}L,$$

which can be written as

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} \left\{ 1 + \widehat{\sigma}^2 \left[\frac{K'(X'X)^{-1}L}{(K'\widehat{\beta})(L'\widehat{\beta})} - \frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^2} \right] \right\}$$

where $1 + \widehat{\sigma}^2 \left[\frac{K'(X'X)^{-1}L}{(K'\widehat{\beta})(L'\widehat{\beta})} - \frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^2} \right]$ can be considered as a correction factor for the estimated ratio estimator.

(iii) The estimate of the variance of the bias-corrected estimator

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{(K'\widehat{\beta})^2}{(L'\widehat{\beta})^2}(A_1 + A_2)$$

where A_1 is the first-order approximation

$$A_{1} = \widehat{\sigma}^{2} \left[\frac{K'(X'X)^{-1}K}{(K'\widehat{\beta})^{2}} - 2 \frac{K'(X'X)^{-1}L}{(K'\beta)(L'\beta)} + \frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^{2}} \right]$$

and A_2 is the additional part from second-order approximation

$$A_{2} = \frac{\widehat{\sigma}^{2} L'(X'X)^{-1} L}{(L'\widehat{\beta})^{2}} \widehat{\sigma}^{4} \left[\frac{K'(X'X)^{-1} K}{(K'\widehat{\beta})^{2}} - 4 \frac{K'(X'X)^{-1} L}{(K'\beta)(L'\beta)} + 2 \frac{L'(X'X)^{-1} L}{(L'\widehat{\beta})^{2}} \right] + \frac{\widehat{\sigma}^{4} (K'(X'X)^{-1} L)^{2}}{(K'\widehat{\beta})^{2} (L'\widehat{\beta})^{2}}$$

Next, we consider the case where the numerator and the denominator of the ratio are independent.

Proposition 4.

(i) If $\hat{\theta}_1$ and $\hat{\theta}_2$ are independent, then the bias for $\hat{\theta}$ becomes

$$\widehat{Bias}(\widehat{\theta}) = \frac{K'\widehat{\beta}}{(L'\widehat{\beta})^3}\widehat{\sigma}^2 L'(X'X)^{-1}L,$$

which can be written as

$$\widehat{Bias}(\widehat{\theta}) = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} \left[\frac{\widehat{\sigma}^2 L'(X'X)^{-1} L}{(L'\widehat{\beta})^2} \right]$$

where $\left[\frac{\widehat{\sigma}^2 L'(X'X)^{-1}L}{(L'\widehat{\beta})^2}\right]$ can be considered as a correction factor for the estimated ratio estimator.

(ii) The bias-corrected estimator $\widehat{\theta}_{BC}$ is given by

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} - \frac{K'\widehat{\beta}}{(L'\widehat{\beta})^3}\widehat{\sigma}^2 L'(X'X)^{-1}L$$

which can be written as

$$\left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2}\right)_{BC} = \frac{K'\widehat{\beta}}{L'\widehat{\beta}} \left[1 - \frac{\widehat{\sigma}^2 L'(X'X)^{-1} L}{(L'\widehat{\beta})^2} \right]$$

where $1 - \frac{\widehat{\sigma}^2 L'(X'X)^{-1} L}{(L'\widehat{\beta})^2}$ can be considered as a correction factor for the estimated ratio estimator.

Econometrics **2025**, 13, 37 17 of 27

(iii) The variance of the bias-corrected estimator

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{(K'\widehat{\beta})^2}{(L'\widehat{\beta})^2}\widehat{\sigma}^2 \left\{ \underbrace{\underbrace{\left[\frac{K'(X'X)^{-1}K}{(K'\widehat{\beta})^2} + \frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^2}\right]}_{first-order\ approximation} + \underbrace{\frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^2}\widehat{\sigma}^2 \left[\frac{K'(X'X)^{-1}K}{(K'\widehat{\beta})^2} + 2\frac{L'(X'X)^{-1}L}{(L'\widehat{\beta})^2}\right]}_{additional\ part\ from\ second-order\ approximation} \right\}$$

We will illustrate this result with an econometric application to show the simplicity of calculation for our method. Let us take the case of the turning point, which has been the subject of numerous economic applications.

6.2. The Turning Point

Consider a classical linear model described by the quadratic regression model

$$y = \beta_0 + \beta_1 x + \beta_2 x^2 + \varepsilon$$

where y is the dependent variable and x the independent variable and ε is an unobserved random error term with expected value $E(\varepsilon)=0$ and variance $V(\varepsilon)=\sigma^2$. A common example of such model is the Kuznets (1955) curve that proposes the relationship between income inequality and income, can be represented by an inverted U shaped curve. Following the Kuznets hypothesis the relation between a country's income equality and economic development is concave, with income equality first increasing and then decreasing as the country's economy is developing. See Bernard et al. (2019), J. Hirschberg and Lye (2005), and Lye and Hirschberg (2018), among others, for the applications and the extensions of this "Kuznets curve". The turning point (or extremum value) is given by

$$\theta = -\frac{\beta_1}{2\beta_2}$$

assuming $\beta_2 \neq 0$, the extremum value θ is a minimum value if $\beta_2 \succ 0$ and a maximum value if $\beta_2 \prec 0$.

In this case,
$$K'=(0,-1,0)$$
 and $L'=(0,0,2)$ and we have $\theta_1=-\beta_1, \quad \theta_1^2=\beta_1^2, \qquad \widehat{V}(\widehat{\theta}_1)=\widehat{V}(\widehat{\beta}_1)=\widehat{\sigma}_{\beta_1}^2$ $\theta_2=2\beta_2, \quad \widehat{\theta}_2^2=4\beta_2^2, \quad \widehat{\theta}_2^3=8\beta_2^3, \quad \widehat{V}(\widehat{\theta}_2)=4\widehat{V}(\widehat{\beta}_2)=4\widehat{\sigma}_{\beta_2}^2$ $\theta_1\theta_2=-2\beta_1\beta_2, \quad \theta_1^2\theta_2^2=4\beta_1^2\beta_2^2, \quad \widehat{Cov}(\widehat{\theta}_1,\widehat{\theta}_2)=2\widehat{Cov}(\widehat{\beta}_1,\widehat{\beta}_2)=-2\widehat{\sigma}_{\widehat{\beta}_1\widehat{\beta}_2}$

In the formulation of the bias for $\hat{\theta}$, the bias-corrected estimator $\hat{\theta}_{BC}$ and its variance, by replacing all these terms, we have the following results:

(i) The bias for $\widehat{\theta}$ is

$$\widehat{Bias}(\widehat{\theta}) = \frac{1}{2} \left[\frac{1}{\widehat{\beta}_2^2} \widehat{\sigma}_{\widehat{\beta}_1 \widehat{\beta}_2} - \frac{\widehat{\beta}_1}{\widehat{\beta}_2^2} \widehat{\sigma}_{\widehat{\beta}_2}^2 \right]$$

which can be written as

$$\widehat{Bias}(\widehat{\theta}) = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} \left(\frac{\widehat{\sigma}_{\widehat{\beta}_2}^2}{\widehat{\beta}_2^2} - \frac{\widehat{\sigma}_{\widehat{\beta}_1 \widehat{\beta}_2}}{\widehat{\beta}_1 \widehat{\beta}_2} \right)$$

where $\begin{pmatrix} \widehat{\sigma}_{\widehat{\beta}_{2}}^{2} - \widehat{\sigma}_{\widehat{\beta}_{1}\widehat{\beta}_{2}} \\ \widehat{\beta}_{1}^{2} - \widehat{\beta}_{1}\widehat{\beta}_{2} \end{pmatrix}$ can be considered as a correction factor to the estimated ratio estimator.

(ii) The bias can be express in terms of the coefficients of variation and the coefficient of co-variation of $\hat{\beta}_1$ and $\hat{\beta}_2$

$$\widehat{\textit{Bias}}(\widehat{\theta}) = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} \left[\left(\frac{1}{t(\widehat{\beta}_2)} \right)^2 - \widehat{\rho} \left(\frac{1}{t(\widehat{\beta}_1)} \right) \left(\frac{1}{t(\widehat{\beta}_2)} \right) \right]$$

where $t(\widehat{\beta}_i)$ denotes the t-statistic for $\widehat{\beta}_i$ for i=1,2, and $\widehat{\rho}$ is the estimate of the correlation coefficient between $\widehat{\beta}_1$ and $\widehat{\beta}_2$ and the term $\left(\frac{1}{t(\widehat{\beta}_2)}\right)^2 - \widehat{\rho}\left(\frac{1}{t(\widehat{\beta}_1)}\right)\left(\frac{1}{t(\widehat{\beta}_2)}\right)$ can be considered as a correction factor to the estimated ratio estimator.

An another alternative form of the bias is

$$\widehat{\mathit{Bias}}(\widehat{\theta}) = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} \left\{ \left(\frac{1}{t(\widehat{\beta}_1)} \right) \left(\frac{1}{t(\widehat{\beta}_2)} \right) \left\lceil \frac{t(\widehat{\beta}_1)}{t(\widehat{\beta}_2)} - \widehat{\rho} \right\rceil \right\}$$

where $\left(\frac{1}{t(\widehat{\beta}_1)}\right)\left(\frac{1}{t(\widehat{\beta}_2)}\right)\left[\frac{t(\widehat{\beta}_1)}{t(\widehat{\beta}_2)}-\widehat{\rho}\right]$ can be considered as a correction factor to the estimated ratio estimator.

(iii) The bias-corrected estimator $\widehat{\theta}_{BC}$

$$\widehat{\theta}_{BC} = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} - \frac{1}{2} \left[\frac{1}{\widehat{\beta}_2^2} \widehat{\sigma}_{\widehat{\beta}_1 \widehat{\beta}_2} - \frac{\widehat{\beta}_1}{\widehat{\beta}_2^3} \widehat{\sigma}_{\widehat{\beta}_2}^2 \right]$$

which can be written as

$$\widehat{\theta}_{BC} = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} \left[1 + \left(\frac{\widehat{\sigma}_{\widehat{\beta}_1 \widehat{\beta}_2}}{\widehat{\beta}_1 \widehat{\beta}_2} - \frac{\widehat{\sigma}_{\widehat{\beta}_2}^2}{\widehat{\beta}_2^2} \right) \right]$$

where $1 + \left(\frac{\widehat{\sigma}_{\widehat{\beta}_1\widehat{\beta}_2}}{\widehat{\beta}_1\widehat{\beta}_2} - \frac{\widehat{\sigma}_{\widehat{\beta}_2}^2}{\widehat{\beta}_2^2}\right)$ can be considered as a correction factor to the estimated ratio estimator

(iv) The bias-corrected estimator $\hat{\theta}_{BC}$ in terms of the coefficient of variation and the coefficient of co-variation of $\hat{\beta}_1$ and $\hat{\beta}_2$ is

$$\widehat{\theta}_{BC} = -\frac{1}{2} \frac{\widehat{\beta}_1}{\widehat{\beta}_2} \left[1 + \left(\widehat{\rho} \left(\frac{1}{t(\widehat{\beta}_1)} \right) \left(\frac{1}{t(\widehat{\beta}_2)} \right) - \frac{1}{t(\widehat{\beta}_2)^2} \right) \right]$$

where $1 + \left(\widehat{\rho}\left(\frac{1}{t(\widehat{\beta}_1)}\right)\left(\frac{1}{t(\widehat{\beta}_2)}\right) - \frac{1}{t(\widehat{\beta}_2)^2}\right)$ can be considered as a correction factor to the estimated ratio estimator.

An another alternative form is

$$\widehat{ heta}_{BC} = -rac{1}{2}rac{\widehat{eta}_1}{\widehat{eta}_2} \left[1 + \left(rac{1}{t(\widehat{eta}_1)}
ight) \left(rac{1}{t(\widehat{eta}_2)}
ight) \left(\widehat{
ho} - rac{t(\widehat{eta}_1)}{t(\widehat{eta}_2)}
ight)
ight]$$

where $1 + \left(\frac{1}{t(\widehat{\beta}_1)}\right) \left(\frac{1}{t(\widehat{\beta}_2)}\right) \left(\widehat{\rho} - \frac{t(\widehat{\beta}_1)}{t(\widehat{\beta}_2)}\right)$ can be considered as a correction factor to the estimated ratio estimator.

(v) The estimate of the variance of the bias-corrected estimator $\widehat{\theta}_{BC}$

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{1}{4} \frac{\widehat{\beta}_{1}^{2}}{\widehat{\beta}_{2}^{2}} \left\{ \begin{array}{c} \underbrace{\left[\frac{\widehat{\sigma}_{\widehat{\beta}_{1}}^{2}}{\widehat{\beta}_{1}^{2}} - 2\frac{\widehat{\sigma}_{\widehat{\beta}_{1}\widehat{\beta}_{2}}}{\widehat{\beta}_{1}\widehat{\beta}_{2}} + \frac{\widehat{\sigma}_{\widehat{\beta}_{2}}^{2}}{\widehat{\beta}_{2}^{2}}\right]}_{\text{first-order approximation}} \\ + \underbrace{\frac{\widehat{\sigma}_{\widehat{\beta}_{2}}^{2}}{\widehat{\beta}_{2}^{2}}}_{\text{additional part from second-order approximation}}^{\widehat{G}_{\widehat{\beta}_{1}\widehat{\beta}_{2}}} + 2\frac{\widehat{\sigma}_{\widehat{\beta}_{1}\widehat{\beta}_{2}}^{2}}{\widehat{\beta}_{2}^{2}}\right] + \underbrace{\frac{(\widehat{\sigma}_{\widehat{\beta}_{1}\widehat{\beta}_{2}})^{2}}{\widehat{\beta}_{1}^{2}\widehat{\beta}_{2}^{2}}}_{\text{additional part from second-order approximation}}^{\widehat{G}_{\widehat{\beta}_{1}\widehat{\beta}_{2}}}$$

(vi) Thus, this variance $\widehat{V}\left[(\widehat{\theta}_{BC})\right]$ can be express in terms of coefficient variation of $\widehat{\beta}_1$ and $\widehat{\beta}_2$ by

$$\widehat{V}\Big[(\widehat{\theta}_{BC})\Big] = \frac{1}{4} \frac{\widehat{\beta}_1^2}{\widehat{\beta}_2^2} \left\{ \begin{array}{c} \underbrace{\left[\widehat{CV}(\widehat{\beta}_1)^2 - 2\widehat{\rho}\widehat{CV}(\widehat{\beta}_1)\widehat{CV}(\widehat{\beta}_2) + \widehat{CV}(\widehat{\beta}_2)^2\right]}_{\text{first-order approximation}} \\ + \underbrace{\widehat{CV}(\widehat{\beta}_2)^2}_{\text{additional part from second-order approximation}} \right\} \\ \\ \underbrace{\left[\widehat{CV}(\widehat{\beta}_1)^2 - 4\widehat{\rho}\widehat{CV}(\widehat{\beta}_1)\widehat{CV}(\widehat{\beta}_2) + \widehat{\rho}^2\widehat{CV}(\widehat{\beta}_1)^2\right]}_{\text{additional part from second-order approximation}} \right\}$$

This variance is easily calculated using the t-statistics for $\hat{\beta}_i$ for i = 1, 2

$$\widehat{V}\left[(\widehat{\theta}_{BC})\right] = \frac{1}{4} \frac{\widehat{\beta}_{1}^{2}}{\widehat{\beta}_{2}^{2}} \left\{ \begin{array}{c} \underbrace{\left[\frac{1}{t(\widehat{\beta}_{1})^{2}} - 2\widehat{\rho}\left(\frac{1}{t(\widehat{\beta}_{1})}\right)\left(\frac{1}{t(\widehat{\beta}_{2})}\right) + \frac{1}{t(\widehat{\beta}_{2})^{2}}\right]}_{\text{first-order approximation}} \\ + \underbrace{\frac{1}{t(\widehat{\beta}_{2})^{2}} \left[\frac{\frac{1}{t(\widehat{\beta}_{1})^{2}} - 4\widehat{\rho}\left(\frac{1}{t(\widehat{\beta}_{1})}\right)\left(\frac{1}{t(\widehat{\beta}_{2})}\right) + \widehat{\rho}^{2}\frac{1}{t(\widehat{\beta}_{1})^{2}}}_{+2\frac{1}{t(\widehat{\beta}_{2})^{2}}} \right]}_{\text{additional part from second-order approximation}} \right\}$$

We have developed a new method for deriving analytical formulae for the bias of the estimator ratios $\hat{\theta}$, the bias-corrected estimator $\hat{\theta}_{BC}$, and the variance of the bias-corrected estimator $\hat{V}(\hat{\theta}_{BC})$. The advantage of this method is that the calculations are quite simple and do not require intensive computations like the bootstrap methods.

7. Simulation Study

7.1. Overview

In this section, we carry out a simulation study to assess the coverage probabilities of the methods presented in the previous section. We also examine, the average length of the confidence intervals. We evaluate the performance of the Fieller interval, the Delta method interval without and with bias correction and the Edgeworth interval without and with bias correction. Let X_1, \ldots, X_n be i.i.d. observations from some distributions F with mean μ_X and variance σ_X^2 , Y_1, \ldots, Y_n be i.i.d. observations from some distributions F with mean F and variance F and F and F and variance F and their ratio F is the correlation coefficient. Let F and F and F and F and F and their ratio F is a consistent estimator of F and F and F and F and F and F and their ratio F is a consistent estimator of F and F are F and F and F are F and F and F are F are F and F are F and F are F and F are F and F are F are F and F are F are F and F are F and F are F and F are F and F are F are F and F are F and F are F are F and F are F are F and F are F and F are F and F are F and F are F are F and F are F and F are F are F and F are F and F are F and F are F are F are F and F are F are F are F are F are F and F are F are F and F are F are F are F are F are F are F and F are F are F are F and F are F are F and F are F are F are F are F and F are F are F and F are F are F are F are F and F are F are F are F are F are F and F are

We generate data from three bivariate distributions: a bivariate normal distribution, and two positively skewed family of distributions. The two families that we consider are the bivariate lognormal distribution and the bivariate mixture ($X_i's$ are lognormal and $Y_j's$ are normal) distribution. We choose three correlation coefficients between X_i and Y_j (-0.8,

Econometrics 2025, 13, 37 20 of 27

0.1, 0.8) and four sample sizes (25, 50, 100, 1000). We use 10,000 data sets. The data are generated as follows:

(a) Bivariate Normal Distribution

$$\begin{pmatrix} X_i \\ Y_i \end{pmatrix} \sim_{i.i.d} N_2 \left(\begin{pmatrix} \mu_X = 7 \\ \mu_Y = 5 \end{pmatrix}, \begin{pmatrix} \sigma_X^2 = 2 & \rho \sigma_X \sigma_Y \\ \rho \sigma_X \sigma_Y & \sigma_Y^2 = 1 \end{pmatrix} \right)$$

(b) Bivariate Mixture Distribution

$$(X_i) = e^{\widetilde{X}_i}$$

$$\left(\begin{array}{c} \widetilde{X}_i \\ Y_i \end{array}\right) \sim_{i.i.d} N_2 \left(\left(\begin{array}{c} \mu_{\widetilde{X}} = 5 \\ \mu_Y = 4 \end{array}\right), \left(\begin{array}{cc} \sigma_{\widetilde{X}}^2 = 0, 2 & \rho \sigma_{\widetilde{X}} \sigma_Y \\ \rho \sigma_{\widetilde{X}} \sigma_Y & \sigma_Y^2 = 0, 5 \end{array}\right)\right)$$

(c) Bivariate Lognormal Distribution

$$\begin{pmatrix} X_i \\ Y_i \end{pmatrix} \sim_{i.i.d} \exp \left\{ N_2 \begin{pmatrix} \mu_X = 5 \\ \mu_Y = 4 \end{pmatrix}, \begin{pmatrix} \sigma_X^2 = 0, 2 & \rho \sigma_X \sigma_Y \\ \rho \sigma_X \sigma_Y & \sigma_Y^2 = 0, 5 \end{pmatrix} \right\}$$

7.2. Results

The results of our simulation are presented in Table 1. The values presented in the table are confidence intervals based on the Fieller method, the Delta method, the Delta method with the bias correction (denoted by Dbc), the Edgeworth method, and the Edgeworth method with the bias correction (denoted by Ebc). The values of the average width (denoted by Width) are the average lengths of the corresponding intervals. For data generated from normal distribution, all intervals give good performance. That is, all coverage probabilities are closer to the nominal level. Average interval lengths (Width) are also comparable for all methods. The Fieller and the Delta confidence intervals are in many cases very close to each other in terms of the coverage probabilities and we can also observe that the average interval lengths for Delta method with the bias correction (Dbc) are less wide than for the Delta method without the bias correction which means that the estimator is more accurate. We also observe that the average interval lengths for the Edgeworth method with the bias correction (Ebc) are narrower than for the Edgeworth method without the bias correction. However, for data generated from the bivariate mixture and bivariate lognormal distributions, Delta methods confidence intervals are obviously inadequate, the coverage probabilities are lower than the nominal level. Fieller's intervals are also insufficient in terms of coverage probabilities. All the other methods give coverage probabilities lower than the nominal level. The Dbc intervals outperform Delta intervals. The Dbc intervals give better coverage probabilities than Delta intervals. They are comparable and sometimes better than the Fieller intervals. Note that the Delta interval has the longest average width whereas the Dbc interval has the shortest average width. The same applies to the Ebc compared to the Edgeworth expansion. We also observe that the Ebc interval performs much better than the Edgeworth interval. This can be explained by the fact that the estimated ratio is biased. Overall, the Edgeworth and the Edgeworth bias corrected appear to be best in terms of coverage probabilities and average width (width). To explore how the correlation coefficients affect the coverage probabilities we performed simulations for different values (-0.8, 0.1, 0.8) from Table 1. The simulation results showed that the correlation coefficients have an impact on the coverage probabilities. The sample sizes have a substantial impact on the coverage probabilities for almost all methods. Among all the methods, the Edgeworth bias-corrected (Ebc) method seems to give a narrower average than the others. The important conclusion from our simulation is that one should use the Edgeworth bias corrected, rather than the Edgeworth expansion. We also consider Econometrics **2025**, 13, 37 21 of 27

other sample sizes and other correlation structures. The results are similar and are not reported here.

In summary, the Edgeworth without and with the bias correction have good performance in terms of coverage probability and average width and should be recommended for constructing confidence intervals when data are from skewed distributions.

Table 1. Coverage probability and average width (Width) of 95% confidence intervals.

ρ	Fieller	Width	Delta	Width	Dbc	Width	Edgeworth	Width	Ebc	Width
(a) Bivariat	e Normal Dis	tribution								
n = 25										
0.8	0.9505	1.9625	0.9491	1.9455	0.9495	1.9364	0.9531	2.0523	0.9532	1.1935
0.1	0.9463	2.0572	0.9452	2.0443	0.9458	2.0115	0.9512	2.0365	0.9511	2.0136
-0.8	0.9485	2.4198	0.9482	2.4041	0.9473	2.1464	0.9528	2.0523	0.9529	2.0310
n = 50										
0.8	0.9505	1.9625	0.9506	1.9485	0.9504	1.9275	0.9526	2.0497	0.9522	1.9210
0.1	0.9489	2.0577	0.9480	2.0443	0.9464	2.0324	0.9510	2.0342	0.9515	2.0387
-0.8	0.9476	2.4187	0.9469	2.4036	0.9478	2.1685	0.9521	2.0415	0.9519	2.0450
n = 100										
0.8	0.9504	1.9753	0.9501	1.9753	0.9503	1.9212	0.9524	2.0520	0.9521	1.9215
0.1	0.9477	2.0678	0.9489	2.0621	0.9463	2.0218	0.9503	2.0365	0.9506	2.0240
-0.8	0.9504	2.3953	0.9468	2.3975	0.9475	2.2358	0.9506	2.0522	0.9505	2.0486
n = 100	0									
0.8	0.9501	1.9780	0.9501	1.9658	0.9500	1.9245	0.9520	2.0568	0.9520	1.9146
0.1	0.9476	2.0749	0.9469	2.0581	0.9460	2.0510	0.9504	2.0412	0.9500	2.0168
-0.8	0.9500	2.3763	0.9470	2.3860	0.9477	2.2045	0.9515	2.0495	0.9514	2.0475
	te Mixture Di									
n = 25	ic mixture Di									
n = 23	0.8286	90.13	0.8214	86.48	0.8297	86.42	0.8674	85.86	0.8815	84.53
0.1	0.8713	107.30	0.8474	103.57	0.8512	103.45	0.8671	102.93	0.8705	102.49
-0.8	0.8970	139.34	0.8570	133.28	0.8980	132.51	0.9013	131.14	0.9051	130.57
	0.057.0	107.01		100.20	0.0700	102.01	0.7010	101111	0.,001	100.07
n = 50	0.8485	91.70	0.8329	88.26	0.8496	87.57	0.8816	87.51	0.8898	86.76
0.0	0.8707	107.18	0.8329	103.78	0.8514	102.12	0.8904	101.45	0.8898	101.14
-0.8	0.8707	107.18	0.8430	132.36	0.8514	132.17	0.8904	130.78	0.9009	101.14
		136.37	0.0333	132.30	0.0390	132.17	0.9002	130.76	0.9121	129.41
n = 100		00.07	0.0610	20.24	0.9737	OF 21	0.0002	96.45	0.0122	07.10
0.8	0.8623	90.87	0.8610	89.24	0.8726	85.21	0.9002	86.45	0.9132	87.10
$0.1 \\ -0.8$	0.8798 0.9015	106.87 137.21	0.8725 0.9104	102.53 130.87	0.8798 0.8805	101.21 130.54	0.9045 0.9068	101.24 130.36	0.9187 0.9208	102.25 129.21
		137.21	0.9104	130.67	0.8803	130.34	0.9008	130.30	0.9208	129.21
n = 100		01.10	0.0505	00.12	0.05/5	04.05	0.0175	00.12	0.0210	04.50
0.8	0.8674	91.10	0.8735	90.13	0.8765	84.25	0.9165	88.12	0.9218	84.59
$0.1 \\ -0.8$	0.8723 0.9001	105.34 136.21	0.8806 0.9312	102.14 131.51	0.8725 0.9422	102.22 130.57	0.9046 0.9185	101.14 130.03	0.9284 0.9298	101.21 129.25
			0.9312	131.31	0.9422	130.37	0.9163	130.03	0.9296	129.23
	e Lognormal	Distribution								
n = 25		. =0.								
0.8	0.8119	1.5936	0.8076	1.4075	0.8121	1.2761	0.8618	1.4063	0.8715	1.2326
0.1	0.9027	2.6677	08546	2.4743	0.9037	2.4106	0.8934	2.4530	0.8963	2.2078
-0.8	0.9232	3.4301	0.8688	3.1422	0.8721	3.1256	0.9066	3.1047	0.9158	3.0985
n = 50	0.0:==	• 0000	0.00=1	4 4-1-	0.0101	4 /	0.00:-	4 455	0.000=	4 4550
0.8	0.8472	2.8002	0.8351	1.4512	0.8486	1.4150	0.8845	1.4526	0.9005	1.4328
0.1	0.9055	2.6200	0.8610	2.4462	0.9065	2.3812	0.8981	2.4175	0.9002	2.4076
-0.8	0.9169	3.0737	0.8688	3.1422	0.8765	3.1027	0.9058	3.1107	0.9084	2.9615
n = 100										
0.8	0.8417	1.1407	0.8427	1.0821	0.8612	1.1835	0.8766	1.0665	0.9106	1.1078
0.1	0.9130	1.8217	0.8819	1.7727	0.9139	1.6941	0.9078	1.7516	0.9178	1.6851
-0.8	0.9244	2.2228	0.8869	2.1603	0.8981	2.2844	0.9134	2.1281	0.9223	2.0675
n = 100										
0.8	0.8626	1.1691	0.8573	1.1126	0.8621	1.1076	0.8938	1.1010	0.9115	1.0981
0.1	0.9088	1.8157	0.8823	1.7375	0.9054	1.6975	0.9057	1.7434	0.9182	1.6896
-0.8	0.9248	2.2106	0.8965	2.2186	0.9045	2.2081	0.9146	2.1126	0.9268	2.0198

Note: Dbc: Delta method with the bias correction; Ebc: Edgeworth method with the bias correction; Width: average confidence interval lenghts; ρ : correlation coefficients.

Econometrics 2025, 13, 37 22 of 27

8. Conclusions

We have developed new methods for constructing confidence intervals for the non-linear functions of parameters In many practical applications, the distribution of the data are not symmetric, in particular when the sample size is small. We propose that the Edgeworth expansion to the statistics makes it possible to remedy this inconvenience. *The Delta method can then be extended using the Edgeworth expansion to obtain a better approximation*. Furthermore, we have shown that the nonlinear functions of the parameters are biased and we have given an analytical expression of the bias of the ratio of the parameters. This has allowed us to define bias-corrected estimators and, more particularly, to calculate the variance associated with these bias-corrected estimators. We have therefore proposed two other new methods: the Delta method with bias correction and the Edgeworth expansion with bias correction. The new methods we propose are straightforward to calculate and do not require intensive calculations such as bootstrapping.

The results of the simulation study showed that our methods generally have better coverage probabilities and confidence width and are narrower than the Delta method and Fieller's method. In the case of bivariate normality, the Delta with bias correction intervals gives better coverage probabilities than the Delta intervals. They are comparable and sometimes better than Fieller's intervals. When the data have been generated from a skewed distribution, the Edgeworth without and with the bias correction have good performance in terms of controlling the coverage probabilities and average length intervals. Thus, we recommend using our new methods with bias correction to construct a reliable confidence interval for nonlinear functions of the estimated parameters.

Finally, it should be noted that the method outlined in this paper for deriving analytical formulae for the bias of ratio estimators, the bias-corrected estimator, and the variance of the bias-corrected estimator can be useful in several econometric and statistical applications, such as, e.g., the long-run elasticities and flexibilities in dynamic models, the willingness to pay value, structural impulse responses, etc.

Funding: This research received no external funding.

Informed Consent Statement: Not applicable.

Data Availability Statement: The original contributions presented in this study are included in the article. Further inquiries can be directed to the corresponding author.

Conflicts of Interest: The author declare no conflicts of interest

Appendix A

The Delta method is useful to approximate the moments of the nonlinear functions of parameters by using Taylor's series expansion. In the literature, only first-order expansions are used to approximate asymptotic sampling distributions. The Delta method provides a compromise to approximate the asymptotic sampling distribution of the ratio parameters $\theta = \theta_1/\theta_2$ where θ_1 and θ_2 are unknwon parameters. However, higher-order expansions are also useful because they can be used to estimate the bias of the ratio parameters and the analytical form of the bias obtained can be used to construct the bias-corrected estimator. We begin with how the variance of the ratio of the parameters in the main text can be approximated with the Delta method. We then extend this approach to obtain the higher-order terms necessary to estimate the bias and derive a bias-corrected estimator.

The variance of a first order Taylor's series expansion,

Let θ is $g(\theta_1, \theta_2) = \theta_1/\theta_2$. On the basis of Taylor's series expansion, the Delta method approximates the variance of a function of estimators of parameters $\hat{\theta} = g(\hat{\theta}_1, \hat{\theta}_2)$ which

Econometrics 2025, 13, 37 23 of 27

estimates $g(\theta_1, \theta_2)$. Since $\widehat{\theta}_1$ and $\widehat{\theta}_2$ are unbiased estimators of θ_1 and θ_2 , respectively, i.e., $E(\widehat{\theta}_i) = \theta_i$ for i = 1, 2, the variance of $\widehat{\theta}$ is

$$V(\widehat{\theta}) = V(g(\widehat{\theta}_1, \widehat{\theta}_2) = G'\Sigma G$$
 (A1)

where *G* is a Jacobian vector containing all the first-order partial derivatives of $g(\widehat{\theta}_1, \widehat{\theta}_2)$ evaluated at θ_i for i = 1, 2.

$$G' = \left[\frac{\partial g(\widehat{\theta}_1, \widehat{\theta}_2)}{\partial \widehat{\theta}_1}, \frac{\partial g(\widehat{\theta}_1, \widehat{\theta}_2)}{\partial \widehat{\theta}_2} \right] = \left[\frac{1}{\theta_2}, \frac{-\theta_1}{\theta_2^2} \right]$$

and Σ is the variance-covariance matrix of $\hat{\theta}_1$ and $\hat{\theta}_2$ defined as follows

$$\Sigma = \left[\begin{array}{cc} V(\widehat{\theta}_1) & Cov(\widehat{\theta}_1\widehat{\theta}_2) \\ Cov(\widehat{\theta}_2\widehat{\theta}_1) & V(\widehat{\theta}_2) \end{array} \right]$$

Solving Equation (A1) and using the estimators $\hat{\theta}_1$ and $\hat{\theta}_2$ to replace for unknown parameters θ_1 and θ_2 , respectively, we get the variance of $\hat{\theta}$

$$V(\widehat{\theta}) = \frac{1}{\widehat{\theta}_2^2} \left[V(\widehat{\theta}_1) - 2 \left(\frac{\widehat{\theta}_1}{\widehat{\theta}_2} \right) Cov(\widehat{\theta}_1, \widehat{\theta}_2) + \left(\frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \right) V(\widehat{\theta}_2) \right]$$

which can be written by

$$V(\widehat{\theta}) = \frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \left[\frac{V(\widehat{\theta}_1)}{\widehat{\theta}_1^2} - 2 \frac{Cov(\widehat{\theta}_1, \widehat{\theta}_2)}{\widehat{\theta}_1 \widehat{\theta}_2} + \frac{V(\widehat{\theta}_2)}{\widehat{\theta}_2^2} \right]$$

Thus, the variance $V(\widehat{\theta})$ can be express in terms of the coefficient of variation and the coefficient of co-variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$

$$\begin{split} V(\widehat{\theta}) &= \frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \Big[CV(\widehat{\theta}_1)^2 - 2CV(\widehat{\theta}_1, \widehat{\theta}_2) + CV(\widehat{\theta}_2)^2 \Big] \\ &= \frac{\widehat{\theta}_1^2}{\widehat{\theta}_2^2} \Big[CV(\widehat{\theta}_1)^2 - 2\rho CV(\widehat{\theta}_1)CV(\widehat{\theta}_2) + CV(\widehat{\theta}_2)^2 \Big] \end{split}$$

where $CV(\widehat{\theta}_i)^2$ is the square of the coefficient of variation of $\widehat{\theta}_i$ for i=1,2 and $CV(\widehat{\theta}_1,\widehat{\theta}_2)$ is the coefficient of co-variation of $\widehat{\theta}_1$ and $\widehat{\theta}_2$ and $\rho=\frac{Cov(\widehat{\theta}_1,\widehat{\theta}_2)}{\left(\sqrt{V(\widehat{\theta}_1)}\right)\left(\sqrt{V(\widehat{\theta}_2)}\right)}$ is the correlation coefficient between $\widehat{\theta}_1$ and $\widehat{\theta}_2$.

Bias of estimator

The first-order Taylor's series approximations may not be accurate in some applications because of bias from truncation of the Taylor's series or small-sample bias in the asymptotic regression parameter variances used in the Taylor's series formulas. A second order Taylor's series expansios of $g(\widehat{\theta}_1,\widehat{\theta}_2)$ is

$$g(\widehat{\theta}_1, \widehat{\theta}_2) = g(\theta_1, \theta_2) + G'\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix} + \frac{1}{2}\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix}' H\begin{pmatrix} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{pmatrix} + R_n \quad (A2)$$

where H is a Hessian matrix containing all the second partial derivatives of $g(\widehat{\theta}_1, \widehat{\theta}_2)$ evaluated at θ_i i = 1, 2.

Econometrics **2025**, 13, 37 24 of 27

$$H = \begin{bmatrix} \frac{\partial^2 g(\widehat{\theta}_1, \widehat{\theta}_2)}{\partial \widehat{\theta}_1^2}, \frac{\partial^2 g(\widehat{\theta}_1, \widehat{\theta}_2)}{\partial \widehat{\theta}_1 \partial \widehat{\theta}_2} \\ \frac{\partial^2 g(\widehat{\theta}_1, \widehat{\theta}_2)}{\partial \widehat{\theta}_2 \partial \widehat{\theta}_1}, \frac{\partial^2 g(\widehat{\theta}_1, \widehat{\theta}_1)}{\partial \widehat{\theta}_2^2} \end{bmatrix} = \begin{bmatrix} 0, -\frac{1}{\theta_2^2} \\ -\frac{1}{\theta_2^2}, \frac{2\theta_1}{\theta_2^3} \end{bmatrix}$$

and the remainder R_n is of order $O\left(\left\|\begin{array}{cc} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{array}\right\|^2\right)$, i.e., $\frac{R_n}{\left\|\begin{array}{cc} \widehat{\theta}_1 - \theta_1 \\ \widehat{\theta}_2 - \theta_2 \end{array}\right\|^2} \longrightarrow 0$ as $\widehat{\theta}_i \longrightarrow \theta_i$

for i = 1, 2 as $n \longrightarrow \infty$.

By taking expectation of Equation (A2) and since $E(\hat{\theta}_i - \theta_i) = 0$ for i = 1, 2, and $E(R_n) = O(n^{-2})$ we obtain

$$E(g(\widehat{\theta}_1, \widehat{\theta}_2)) = g(\theta_1, \theta_2) + \frac{1}{2}tr\{H\Sigma\} + O(n^{-2})$$
$$E(\widehat{\theta}) = \theta + \frac{1}{2}tr\{H\Sigma\} + O(n^{-2})$$

where tr(.) denotes the trace of matrix, then the bias for $\hat{\theta}$ is defined by

$$Bias(\widehat{\theta}) = E(\widehat{\theta}) - \theta = \frac{1}{2}tr\{H\Sigma\} + O(n^{-2}) = \frac{1}{2}(vecH)'vec\Sigma + O(n^{-2})$$

where vec(.) denotes the vectorisation operator which stacks the columns of the matrix and the matrix H is symmetric so that vecH' = vecH.

Since H and Σ are unknown, we estimate bias as

$$\widehat{Bias}(\widehat{\theta}) = \frac{1}{2} tr \Big(\widehat{H} \widehat{\Sigma} \Big) + O(n^{-2} = \frac{1}{2} (vec \widehat{H})' vec \widehat{\Sigma} + O(n^{-2})$$

where \widehat{H} is the estimate of the Hessian matrix of the second-order partial derivatives and $\widehat{\Sigma}$ is the estimate of the variance–covariance matrix of $\widehat{\theta}_1$ and $\widehat{\theta}_2$.

This yields

$$\widehat{Bias}(\widehat{\theta}) = -\frac{1}{\widehat{\theta}_2^2} \widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2) + \frac{\widehat{\theta}_1}{\widehat{\theta}_2^3} \widehat{V}(\widehat{\theta}_2) + O(n^{-2})$$

which can be written as

$$\widehat{Bias}(\widehat{\theta}) = \frac{\widehat{\theta}_1}{\widehat{\theta}_2} \left[\frac{\widehat{V}(\widehat{\theta}_2)}{\widehat{\theta}_2^2} - \frac{\widehat{Cov}(\widehat{\theta}_1, \widehat{\theta}_2)}{\widehat{\theta}_1 \widehat{\theta}_2} \right] + O(n^{-2})$$

The approximation of the variance of $\hat{\theta}$ with a second-order term

The calculation of the variance of the second order Taylor series reveals the covariances between the random variables and gives a better approximation.

To facilitate notation, let us define the random vector $z = (\widehat{\theta}_1 - \theta_1, \widehat{\theta}_2 - \theta_2)'$ with E(z) = 0, $E(zz') = \Sigma$ and z is a normal random variable $z \sim N(0, \Sigma)$.

We can rewrite the second order of Taylor's expansion as follows

$$g(\widehat{\theta}_1, \widehat{\theta}_2) = g(\theta_1, \theta_2) + G'z + \frac{1}{2}z'Hz + R_n$$

and its variance is

$$V(g(\widehat{\theta}_1,\widehat{\theta}_2)) = V(G'z) + \frac{1}{4}V(z'Hz) + Cov(G'z,z'Hz)$$

To obtain the variance $V(g(\widehat{\theta}_1, \widehat{\theta}_2))$ we need to calculate the three terms

(i)
$$V(G'z) = G'\Sigma G;$$

Econometrics **2025**, 13, 37 25 of 27

(ii)
$$\frac{1}{4}V(z'Hz) = \frac{1}{4} \Big\{ E[z'Hz]^2 - [E(z'Hz)]^2 \Big\}$$

$$= \frac{1}{4} \Big\{ [tr(H\Sigma)]^2 + 2tr(H\Sigma)^2 - [tr(H\Sigma)]^2 \Big\}$$

$$= \frac{1}{2}tr(H\Sigma)^2 ;$$
(iii)
$$Cov(G'z, z'Hz) = G'E[zz'Hz]$$

$$= G'E[z \otimes zz']'vecH$$

$$= 0.$$

Since odd moments of z are zero. Thus, the linear form G'z and the quadratic form z'Hz are uncorrelated.

By combining these three results, we obtain the following result

$$V\left[g(\widehat{\theta}_{1},\widehat{\theta}_{2})\right] = \underbrace{\mathcal{G}'\Sigma\mathcal{G}}_{\text{first-order part}} + \underbrace{\frac{1}{2}tr\left[(H\Sigma)^{2}\right]}_{\text{second-order part}}$$

$$= \underbrace{\mathcal{G}'\Sigma\mathcal{G}}_{\text{first-order part}} + \underbrace{\frac{1}{2}(vecH)'(\Sigma\otimes\Sigma)vecH}_{\text{second-oder part}}$$

where the first order part $G'\Sigma G$ is the variance of $\widehat{\theta}$ corresponding to a first order approximation and the second order part corresponding to an additional part from second-order approximation which permit to take into account the correlation between the random variables.

Let
$$\widehat{V}\left[g(\widehat{\theta}_1,\widehat{\theta}_2)\right]$$
 be the estimate of the variance $V\left[g(\widehat{\theta}_1,\widehat{\theta}_2)\right]$ defined by

$$\widehat{V}\left[g(\widehat{\theta}_{1},\widehat{\theta}_{2})\right] = \underbrace{\widehat{G}'\widehat{\Sigma}\widehat{G}}_{\text{first-order part}} + \underbrace{\frac{1}{2}(vec\widehat{H})'(\widehat{\Sigma}\otimes\widehat{\Sigma})vec\widehat{H}}_{\text{second-oder part}}$$

This yields

$$\begin{split} \widehat{V}\left[g(\widehat{\theta}_{1},\widehat{\theta}_{2})\right] &= \underbrace{\frac{1}{\widehat{\theta}_{2}^{2}} \left[\widehat{V}(\widehat{\theta}_{1}) - 2\left(\frac{\widehat{\theta}_{1}}{\widehat{\theta}_{2}}\right) \widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2}) + \left(\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}\right) \widehat{V}(\widehat{\theta}_{2})\right]}_{\text{first-order approximation}} \\ &+ \underbrace{\frac{1}{\widehat{\theta}_{2}^{4}} \widehat{V}(\widehat{\theta}_{2}) \left[\widehat{V}(\widehat{\theta}_{1}) - 4\left(\frac{\widehat{\theta}_{1}}{\widehat{\theta}_{2}}\right) \widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2}) + 2\left(\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}\right) \widehat{V}(\widehat{\theta}_{2})\right]}_{+\frac{1}{\widehat{\theta}_{2}^{4}} \widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})^{2}} \end{split}$$

additional part from second-order approximation

which can be written as

$$\begin{split} \widehat{V}\left[g(\widehat{\theta}_{1},\widehat{\theta}_{2})\right] &= \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}} \left[\underbrace{\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} - 2\frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})}{\widehat{\theta}_{1}\widehat{\theta}_{2}} + \underbrace{\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}} \right]_{\text{first-order approximation}} \\ &+ \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}} \left\{ \begin{array}{c} \underbrace{\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}}} \left[\frac{\widehat{V}(\widehat{\theta}_{1})}{\widehat{\theta}_{1}^{2}} - 4\frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})}{\widehat{\theta}_{1}\widehat{\theta}_{2}} + 2\frac{\widehat{V}(\widehat{\theta}_{2})}{\widehat{\theta}_{2}^{2}} \right] \\ &+ \underbrace{\frac{\widehat{Cov}(\widehat{\theta}_{1},\widehat{\theta}_{2})^{2}}{\widehat{\theta}_{1}^{2}\widehat{\theta}_{2}^{2}}} \end{array} \right] \end{split}$$

additional part from second-order approximation

Thus, this variance can be express in terms of the coefficient of variation and the coefficient of co-variation of $\hat{\theta}_1$ and $\hat{\theta}_2$.

Econometrics **2025**, 13, 37 26 of 27

$$\begin{split} \widehat{V}\left[g(\widehat{\theta}_{1},\widehat{\theta}_{2})\right] &= \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \\ &+ \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \\ &+ \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \underbrace{\left[\widehat{CV}(\widehat{\theta}_{1})^{2} - 4\widehat{CV}(\widehat{\theta}_{1},\widehat{\theta}_{2}) + 2\widehat{CV}(\widehat{\theta}_{2})^{2}\right]}_{\text{additional part from second-order approximation}} \\ &= \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \\ &+ \underbrace{\frac{\widehat{\theta}_{1}^{2}}{\widehat{\theta}_{2}^{2}}}_{\text{first-order approximation}} \\ \\ &+ \underbrace{\frac{\widehat{\theta}_{1}^{$$

additional part from second-order approximation

References

Andrews, I., Stock, J. H., & Sun, L. (2019). Weak instruments in instrumental variables regression: Theory and practice. *Annual Review of Economics*, 11(1), 727–753. [CrossRef]

Beale, E. M. L. (1962). Some use of computers in operational research. Industrielle Organisation, 31, 27-28.

Bernard, J.-T., Chu, B., Khalaf, L., & Voia, M. (2019). Non-standard confidence sets for ratios and tipping point with applications to dynamic panel data. *Annals of Economics and Statistics*, 134, 79–108. [CrossRef]

Bernard, J.-T., Idoudi, N., Khalaf, L., & Yélou, C. (2007). Finite sample inference methods for dynamic energy demand models. *Journal of Applied Econometrics*, 22, 1211–1226. [CrossRef]

Beyene, J., & Moineddin, R. (2005). Methods for confidence interval estimation of a ratio parameter with application to location quotients. *BMC Medical Research Methodology*, 5, 32. [CrossRef] [PubMed]

Briggs, A., & Fenn, P. (1998). Confidence intervals or surfaces? Uncertainty on the cost-effectiveness plane. *Health Economics*, 7, 723–740. [CrossRef]

Cochran, W. G. (1977). Sampling techniques. John Wiley & Sons.

Dalabehera, M., & Sahoo, L. N. (1995). Efficiencies of six almost unbiased ratio estimators under a particular model. *Statistical Papers*, 36, 61–67. [CrossRef]

David, I. P., & Sukhatme, B. V. (1974). On the bias and mean square error of the ratio estimator. *Journal of the American Statistical Association*, 69(346), 464–466. [CrossRef]

Dorfman, J. H., Kling, C. L., & Sexton, R. J. (1990). Confidence intervals for elasticities and flexibilities: Reevaluating the ratios of normals case. *American Journal of Agricultural Economics*, 72, 1006–1017. [CrossRef]

Dufour, J.-M. (1997). Some impossibility theorems in econometrics with applications to structural and dynamic models. *Econometrica*, 65(6), 1365–1387. [CrossRef]

Dufour, J.-M., Flachaire, E., Khalaf, L., & Zalghout, A. (2018). Confidence sets for inequality measures: Fieller-type methods. In W. Green, L. Khalaf, P. Makdissi, R. Sickles, M. Veall, & M. Voia (Eds.), *Productivity and inequality*. Springer.

Dufour, J.-M., Flachaire, E., Khalaf, L., & Zalghout, A. (2024). Identification-robust methods for comparing inequality with an application to regional disparities. *The Journal of Economic Inequality*, 22, 433–452. [CrossRef]

Faraggi, D., Izikson, P., & Reiser, B. (2003). Confidence intervals for the 50 per cent response dose. *Statistics in Medicine*, 22, 1977–1988. [CrossRef]

Fieller, E. C. (1954). Some problems in interval estimation. Journal of the Royal Statistical Society, Series B, 16, 175–185. [CrossRef]

Hall, P. (1992a). On the removal of skewness by transformation. *Journal of the Royal Statistical Society, Series B*, 54, 221–228. [CrossRef] Hall, P. (1992b). *The bootstrap and edgeworth expansion*. Springer

Hayya, J., Armstrong, D., & Gressis, N. (1975). A note on the ratio of two normall distributed variables. *Management Sciences*, 21, 1338–1341. [CrossRef]

Hirschberg, J., & Lye, J. (2005). *Inferences for the extremm of quadratic regression models*. Technical report. 906. Department of Economics, The University of Melbourne.

Hirschberg, J., & Lye, J. (2010a). A geometric comparison of the delta and fieller confidence intervals. *The American Statistician*, 64(3), 234–241. [CrossRef]

Hirschberg, J., & Lye, J. (2010b). A reinterpretation of interactions in regressions. Applied Economics Letters, 17, 427–430. [CrossRef]

Econometrics **2025**, 13, 37 27 of 27

Hirschberg, J., & Lye, J. (2010c). Two geometric representations of confidence intervals for ratios of linear combinations of regression parameters: An application of the NAIRU. *Economics Letters*, 108, 73–76. [CrossRef]

- Hirschberg, J., & Lye, J. (2017). Inverting the indirect—The ellipse and the boomerang: Visualizing the confidence intervals of the structural coefficient from two-stage least squares. *Journal of Econometrics*, 199, 173–183. [CrossRef]
- Hirschberg, J. G., Lye, J. N., & Slottje, D. J. (2008). Inferential methods for elasticity estimates. *Journal of Econometrics*, 147(2), 299–315. [CrossRef]
- Hwang, J. (2019). *Note on edgeworth expansions and asymptotic refinements of percentile t-bootstrap methods*. Working paper. University of Connecticut.
- Krinsky, I., & Robb, A. L. (1986). On approximating the statistical properties of elasticities. *The Review of Economics and Statistics*, 68, 715–719. [CrossRef]
- Kuznets, S. (1955). International differences in capital formation and financing. In *Capital formation and economic growth* (pp. 19–111). Princeton University Press.
- Leitner, L. (2024). Imprecision in the estimation of willingness to pay using subjective well-being data. *Journal of Happiness Studies*, 25(7), 94. [CrossRef]
- Li, H., & Maddala, G. S. (1999). Bootstrap variance estimation of nonlinnear functions of parameters: An application to long-run elasticities of energy demand. *The Review of Economics and Statistics*, 81(4), 728–733. [CrossRef]
- Lye, J., & Hirschberg, J. (2018). Ratios of parameters: Some econometric examples. *Australian Economic Review*, 51(4), 578–602. [CrossRef] Olea, J. L. M., Stock, J. H., Watson, M. W. (2021). Inference in structural vector autoregressions identified with an external instrument. *Journal of Econometrics*, 225(1), 74–87. [CrossRef]
- Parr, W. C. (1983). A note on the jacknife the bootstrap and te deltta method estimators of bias and variance. *Biometrika*, 70(3), 719–722. [CrossRef]
- Raghav, Y. S., Ahmadini, A. A. H., Mahnashi, A. M., & Rather, K. U. I. (2025). Enhancing estimation efficiency with proposed estimator: A comparative analysis of poisson regression-based mean estimators. *Kuwait Journal of Science*, 52(1), 100282. [CrossRef]
- Scheffé , H. (1970). Multiple testing versus multiple estimation. Improper confidence sets. Estimation of directions and ratios. *The Annals of Mathematical Statistics*, 41(1), 1–29. [CrossRef]
- Sitter, R. R., & Wu, C. F. J. (1993). On the accurracy of Fieller intervals for binar response data. *Journal of the American Statistical Association*, 88, 1021–1025. [CrossRef]
- Staiger, D., Stock, J., & Watson, M. (1997). The nairu, unemployment and monetary policy. *Journal of Economic Perspectives*, 11, 33–49. [CrossRef]
- Swain, A. K. P. C., & Dash, P. (2020). On a class of almost unbiased ratio type estimators. *Journal of Statistical Theory and Applications*, 19(1), 28–35. [CrossRef]
- Tin, M. (1965). Comparison of some ratio estimators. Journal of the American Statistical Association, 60(309), 294–307. [CrossRef]
- von Luxburg, U., & Franz, V. H. (2009). A geometric approach to confidence sets for ratios: Fieller's theorem, generalizations and bootstrap. *Statistica Sinica*, 19, 1095–1117.
- Wang, P., Xu, S., Wang, Y. X., Wu, B., Fung, W. K., Gao, G., & Liu, N. (2021). Penalized Fieller's confidence interval for the ratio of bivariate normal means. *Biometrics*, 77(4), 1355–1368. [CrossRef] [PubMed]
- Wang, Y., Wang, S., & Carroll, R. J. (2015). The direct integral method for confidence intervals for the ratio of two location parameters. *Biometrics*, 71, 704–713. [CrossRef] [PubMed]
- Woglom, G. (2001). More results on the exact small sample properties of the instrumental variable estimator. *Econometrica*, 69(5), 1381–1389. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.